

Diverging Ozone Trends Above Western North America: Boundary Layer Decreases Versus Free Tropospheric Increases

Kai-lan Chang, Owen Cooper, Gavin Rodriguez, Laura Iraci, Emma Yates, Matthew Johnson, Audrey Gaudel, Daniel Jaffe, Noah Bernays, Hannah

Clark, et al.

► To cite this version:

Kai-lan Chang, Owen Cooper, Gavin Rodriguez, Laura Iraci, Emma Yates, et al.. Diverging Ozone Trends Above Western North America: Boundary Layer Decreases Versus Free Tropospheric Increases. Journal of Geophysical Research: Atmospheres, 2023, 128 (8), 10.1029/2022JD038090. hal-04789610

HAL Id: hal-04789610 https://hal.science/hal-04789610v1

Submitted on 19 Nov 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



JGR Atmospheres

RESEARCH ARTICLE

10.1029/2022JD038090

Key Points:

- The median free tropospheric trend above western North America (1995–2021) remained positive despite the COVID-19 economic downturn in 2020
- Stronger positive ozone trends in all seasons and nearly all percentiles are found above the California subdomain
- In contrast ozone is decreasing at rural surface sites, especially at the 95th percentile

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

K.-L. Chang, kai-lan.chang@noaa.gov

Citation:

Chang, K.-L., Cooper, O. R., Rodriguez, G., Iraci, L. T., Yates, E. L., Johnson, M. S., et al. (2023). Diverging ozone trends above western North America: Boundary layer decreases versus free tropospheric increases. *Journal of Geophysical Research: Atmospheres*, *128*, e2022JD038090. https://doi. org/10.1029/2022JD038090

Received 26 OCT 2022 Accepted 28 MAR 2023

Author Contributions:

Conceptualization: Kai-Lan Chang, Owen R. Cooper, Laura T. Iraci, Matthew S. Johnson

Data curation: Laura T. Iraci, Emma L. Yates, Matthew S. Johnson, Audrey Gaudel, Daniel A. Jaffe, Noah Bernays, Hannah Clark, Peter Effertz, Thierry Leblanc, Irina Petropavlovskikh, Bastien Sauvage, David W. Tarasick Formal analysis: Kai-Lan Chang, Gavin

Rodriguez Funding acquisition: Kai-Lan Chang, Owen R. Cooper, Laura T. Iraci, Emma L. Yates, Matthew S. Johnson

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Diverging Ozone Trends Above Western North America: Boundary Layer Decreases Versus Free Tropospheric Increases

Kai-Lan Chang^{1,2}, Owen R. Cooper^{1,2}, Gavin Rodriguez^{1,2}, Laura T. Iraci³, Emma L. Yates^{3,4}, Matthew S. Johnson³, Audrey Gaudel^{1,2}, D, Daniel A. Jaffe^{5,6}, Noah Bernays⁵, Hannah Clark⁷, Peter Effertz^{1,8}, Thierry Leblanc⁹, Irina Petropavlovskikh^{1,8}, Bastien Sauvage¹⁰, and David W. Tarasick¹¹

¹Cooperative Institute for Research in Environmental Sciences, University of Colorado Boulder, Boulder, CO, USA, ²NOAA Chemical Sciences Laboratory, Boulder, CO, USA, ³NASA Ames Research Center, Earth Science Division, Moffett Field, CA, USA, ⁴Bay Area Environmental Research Institute, Moffett Field, CA, USA, ⁵School of STEM, University of Washington Bothell, Bothell, WA, USA, ⁶Department of Atmospheric Sciences, University of Washington Seattle, Seattle, WA, USA, ⁷IAGOS-AISBL, Brussels, Belgium, ⁸NOAA Global Monitoring Laboratory, Boulder, CO, USA, ⁹Jet Propulsion Laboratory, California Institute of Technology, Wrightwood, CA, USA, ¹⁰Laboratoire d'Aérologie (LAERO), Université Toulouse III - Paul Sabatier, CNRS, Toulouse, France, ¹¹Environment and Climate Change Canada, Toronto, ON, Canada

Abstract This study has produced an improved percentile and seasonal (median) trend estimate of free tropospheric ozone above western North America (WNA), through a data fusion of ozonesonde, lidar, commercial aircraft, and field campaign measurements. Our method combines heterogeneous data sets according to the consensus data characteristics and inherent uncertainty in order to produce our best fused product. In response to different data collection environments (in situ or ground-based), we investigate the ozone variability based on a wide range of percentiles, which is preferable for trend detection due to tropospheric ozone's high degree of heteroscedasticity (i.e., inconsistent trends and variability between different ozone percentiles). We then compare the ozone trends and variability above the California sub-domain to the full WNA region for better understanding of the correlations between different regional scales. In California, the 1995–2021 percentile (from the 5th to 95th) and seasonal trends are clearly positive in terms of high signal-to-noise ratios. The magnitude of the trends is generally weaker over WNA compared to California, but reliable positive trends can still be found between the 10th and 70th percentiles, as well as winter and summer, whereas autumn shows a negative trend over the same period. In addition, dozens of rural surface sites across the region are selected to represent the boundary layer variability. In contrast to increasing free tropospheric ozone, we find overall strong negative surface trends since 1995, with the greatest divergence found in summer. Throughout the analysis implications of the COVID-19 economic downturn on ozone variability are discussed.

Plain Language Summary Free tropospheric ozone above western North America has increased since the mid-1990s. Despite an observed drop of ozone in 2020 due to the COVID-19 economic downturn, this observation-based study shows the overall free tropospheric ozone trends have not been offset and continued to increase over 1995–2021, mainly driven by strong positive trends in winter and summer. In combination with the strong negative trends observed at rural surface sites over the same period, this study adds to the growing body of evidence that surface trends are frequently disconnected from the general increases observed in the free troposphere.

1. Introduction

Tropospheric ozone is a major greenhouse gas (Gulev et al., 2021), and a pollutant harmful to human health (Brauer et al., 2016; Fleming et al., 2018; Malashock et al., 2022) and crop productivity (Emberson et al., 2000; Mills et al., 2011, 2018). From a climate change perspective, ozone is the third most important greenhouse gas (after CO_2 and CH_4) and its radiative forcing is proportional to the global tropospheric ozone burden, which contributes to global warming on relatively long timescales of decades (see Chapter 7 of IPCC AR6, 2021). Therefore, long-term ozone trends at remote sites and in the free troposphere are typically based on observations from all months of the year, in addition to seasonal-based analyses (Gulev et al., 2021; Oltmans et al., 2006, 2013; Szopa et al., 2021). The Intergovernmental Panel on Climate Change (IPCC) sixth assessment





Investigation: Kai-Lan Chang, Owen R. Cooper, Laura T. Iraci, Matthew S. Johnson

Methodology: Kai-Lan Chang, Owen R. Cooper

Project Administration: Owen R. Cooper, Laura T. Iraci Resources: Kai-Lan Chang, Owen R.

Cooper Software: Kai-Lan Chang, Gavin

Rodriguez Supervision: Owen R. Cooper, Laura

T. Iraci

Validation: Kai-Lan Chang Visualization: Kai-Lan Chang, Owen R. Cooper

Writing – original draft: Kai-Lan Chang, Owen R. Cooper, Laura T. Iraci, Matthew S. Johnson, Daniel A. Jaffe, Noah Bernays, Hannah Clark, Peter Effertz, Thierry Leblanc, Irina Petropavlovskikh, Bastien Sauvage, David W Tarasick

Writing – review & editing: Kai-Lan Chang, Owen R. Cooper, Laura T. Iraci, Matthew S. Johnson, Daniel A. Jaffe, Hannah Clark, Thierry Leblanc, Irina Petropavlovskikh, David W. Tarasick report (AR6; Gulev et al., 2021; Szopa et al., 2021) assessed output from an ensemble of chemistry climate models and concluded that the tropospheric ozone burden increased by 45% from 1850 to the present day, mainly due to the burning of fossil fuels. This modeled increase is consistent with limited observations (surface and free tropospheric) from the mid-20th century through the early 21st century (Tarasick et al., 2019). The IPCC AR6 report also concluded with high confidence that observed free tropospheric ozone has increased at northern mid-latitudes and across the tropics since the mid-1990s. A combined product from NASA's OMI (Ozone Monitoring Instrument) and MLS (Microwave Limb Sounder) satellite instruments shows positive trends of the tropospheric column ozone across the northern tropics and northern mid-latitudes over 2004–2021 (Dunn et al., 2022; Ziemke et al., 2022), indicating that the broad increases of tropospheric ozone are continuing to the present day.

Meanwhile, the COVID-19 (SARS-CoV-2) pandemic emerged in late 2019 and triggered a worldwide economic downturn in 2020, which reduced emissions of ozone precursors. Studies have shown a weak but consistent ozone drop in the mid-troposphere across the Northern Hemisphere in response to the emissions reduction in 2020 (Bouarar et al., 2021; Chang et al., 2022; Clark et al., 2021; Miyazaki et al., 2021; Steinbrecht et al., 2021). Specifically, Steinbrecht et al. (2021) and Ziemke et al. (2022) observed approximately 7% or 4 ppbv ozone reductions at multiple ozone monitoring sites within the 1-8 km column during April-August 2020. By integrating ozonesonde, lidar and commercial aircraft data, Chang et al. (2022) quantified an overall free tropospheric ozone reduction of 3.6 [±1.8] ppbv above Europe and 2.8 [±1.9] ppbv above western North America (WNA) in 2020 with respect to 1994–2019. The diminished ozone levels in 2020 slightly weakened the positive free tropospheric ozone trends observed for the period 1994-2019. In terms of model studies, a 5%-15% ozone reduction was estimated for spring and summer of 2020 above the extratropical Northern Hemisphere (Bouarar et al., 2021). Miyazaki et al. (2021) estimated an ozone decrease of 5 ppby at the surface and 3 ppby in the mid-troposphere (500 hPa) above northern mid-latitudes in spring of 2020. Compared to the same period in March-June 2019, Miyazaki et al. (2021) also show that free tropospheric ozone (700 hPa) reductions can be observed from the CrIS (Cross-Track Infrared Sounder) satellite instrument, with one of the most remarkable reductions (~10 ppb) found in June above the western USA. By including an additional year of data in 2021, this study will examine if the COVID-19 economic downturn has had a lasting impact on the long-term free tropospheric ozone trends.

At the surface, several studies found that ozone increased broadly in many urban areas during the lockdown period (Cooper et al., 2021; Gkatzelis et al., 2021; Keller et al., 2021; Sokhi et al., 2021), likely because reduced NO emissions tend to limit the ozone destruction in highly polluted urban areas (Sillman, 1999). Jaffe et al. (2022) analyzed surface data from 22 urban monitoring sites in the U.S. and found that COVID emission reductions significantly reduced peak ozone in the Eastern U.S., but in the western U.S. peak ozone increased in 2020 due to wildfire emissions. While those surface studies focused on urban areas, this study will investigate the COVID-19 impact on surface ozone trends and variability at rural sites across WNA.

This study describes a refined methodology for exploring trends across the full range of the ozone distribution (e.g., from the 5th percentile to the 95th percentile). Conventionally, heteroscedasticity in regression analvsis refers to data variability that is not constant and varies over time. If data are homogeneous, all percentile trends should be similar. Given that previous studies generally found diverse tropospheric ozone percentile trends (Cooper et al., 2010; Gaudel et al., 2020), heteroscedasticity is an expected characteristic of ozone trend analysis. Previous investigations of ozone percentile trends have typically calculated the sample percentile at each time interval and then applied the mean or median-based trend detection technique to the percentile time series (Cooper et al., 2012; Lin et al., 2017; Michael et al., 2019; Strode et al., 2015). However, that approach implicitly assumes that the data samples are abundant enough to derive a representative sample percentile for each given month. Therefore, such an approach to study the trends in extreme events is usually limited to surface observations measured at a regular and high frequency (e.g., hourly data), instead of sparsely sampled free tropospheric ozone profiles. An alternative approach to evaluate the percentile trends is through a well-developed technique known as quantile regression (Koenker & Hallock, 2001). This method is designed to detect heterogeneous distributional changes, and has been applied to the study of surface ozone extreme events (Baur et al., 2004; Munir et al., 2012; Porter et al., 2015), and to explore extreme percentile variability from commercial aircraft data (Gaudel et al., 2020).

Whereas these two approaches discussed above have been used to successfully evaluate ozone percentile trends, the rationale behind them is slightly different. For example, if we aim to derive the 95th percentile change of an



ozone time series based on daily values, the difference between these two approaches to derive the percentile trends can be understood as, respectively:

- 1. A 2-step method: the 95th sample percentile is first calculated from daily values for each individual month, and then the trend is fit to the resulting monthly time series (i.e., focusing on targeted sample percentiles).
- 2. A single-step method: quantile regression is directly applied to the full daily time series for deriving the 95th percentile trend, as it is designed for investigating distributional change (i.e., based on all available data samples).

Even though these two approaches often arrive at similar conclusions, quantile regression provides a more stable estimation when data sample sizes are deemed to be less sufficient, since all data points contribute to the estimation, rather than merely a portion of data (Krock et al., 2022). Nevertheless, the discussion so far only considers ozone time series collected from a single source (e.g., a single monitoring station). If an assessment of trends is based on multiple data sources, the complications associated with varying data characteristics should be taken into account in advance, such as spatial variability (Chang et al., 2017; Wood et al., 2017) or various measurement platforms with differing sampling frequencies (Chang et al., 2022).

This paper improves upon previous estimates of the free tropospheric ozone trends above WNA (Chang et al., 2022; Cooper et al., 2010; Lin et al., 2015); through a more detailed investigation of seasonal and percentile ozone trends based on a data fusion of multiple sources of data platforms (Chang et al., 2022). The purpose of data fusion is to integrate multiple heterogeneous data sets to produce more consistent and accurate information. We highlight the benefit of an integration of all available data sources, which can increase our ability to detect trends in extreme percentiles for both the large domain of WNA and the much smaller sub-region of California. Further evidence of increasing free tropospheric ozone is provided by investigating ozone seasonal and percentile variability at the Mt. Bachelor Observatory, where the nighttime data are considered to be representative of the lower free troposphere, and a strong ozone increase has been previously reported (Gratz et al., 2015). An additional analysis of surface trends at rural sites across the western United States is included to explore the differences in ozone trends between the free troposphere and the boundary layer.

Section 2 outlines the sources of free tropospheric ozone profiles and rural surface measurements used in this study, and the statistical methodology for data fusion and percentile trend estimation. The results of trend analyses for the free tropospheric ozone and the rural surface ozone are provided in Sections 3 and 4, respectively. A summary of this study is provided in Section 5.

2. Data and Method

2.1. Data

The free tropospheric ozone measurements used in this study include ozonesonde, lidar, commercial aircraft, and field campaign observations. We focus on the period 1995–2021 when the data density is great enough to calculate regional-scale trends. Since different data sets report ozone values at different resolutions, measurements from each profile are aggregated into a vertical resolution of 10 hPa over 700–300 hPa, so that all data sets are aligned to the same vertical coordinate. A map view of flight tracks and data densities from the IAGOS (In-Service Aircraft for a Global Observing System) program and AJAX (Alpha Jet Atmospheric eXperiment) project, and the locations of ozonesonde and lidar stations are shown in Figure 1.

The data sets are as follows: (a) ozonesondes launched at Boulder (Colorado, 1967–2021), Trinidad Head (THD, California, 1997–2021), Edmonton (Canada, 1970–2021), Kelowna (Canada, 2003–2017) and Port Hardy (Canada, 2018–2021), with a sampling frequency of approximately once per week (Sterling et al., 2018; Tarasick et al., 2016); (b) The lidar operated at the Jet Propulsion Laboratory Table Mountain Facility (TMF, California) has typically measured 2–5 profiles per week since 1999 (McDermid et al., 2002), with an increased frequency in the most recent years (Chouza et al., 2019); (c) The IAGOS program has measured ozone from multiple commercial aircraft since 1994 (Blot et al., 2021; Nédélec et al., 2015; Petzold et al., 2015). Ozone profiles above WNA mainly occurred above the international airports at Vancouver, San Francisco, Portland, and Los Angeles; a limitation of the IAGOS data set is the occurrence of many intermittent data gaps over this region (see Figure 1 for the annual sample size); and (d) the NASA AJAX field campaign data set includes 199 flights during 2011–2018 (Iraci et al., 2021; Yates et al., 2016, 2017), and the SNAAX (Scientific Aviation NASA Ames Airborne





Figure 1. A map view of flight tracks and data densities from the aircraft measurements above western North America (1994–2021), and the locations of ozonesonde (circles) and lidar (square) stations. Also shown are numbers of annual profiles from different data platforms (colored by the range of sample sizes), where (*S*) represents the ozonesonde station, (*L*) represents lidar station, and (all) represents the regional sum.

eXperiment) data set includes 10 flights during 2018–2019. While most of these data sets were previously used in the recent study by Chang et al. (2022) we have been able to increase the size of the overall data set with the inclusion of the AJAX and SNAAX profiles, and additional IAGOS profiles from the southwestern USA: Denver (2), Fresno (2), Las Vegas (90), Los Angeles (404), Phoenix (24), San Jose (2) and San Diego (120). These additional 853 profiles increased the size of the overall WNA data set (10,403 total profiles) by 9%, and allowed the creation of a new aircraft ozone time series for the southwestern USA (mainly California), when merged with the 437 IAGOS profiles above San Francisco (previously used by Chang et al., 2022 in combination with IAGOS profiles above Portland, Seattle, and Vancouver). In addition, this analysis includes one additional year (2021) of data from all sites, if available, compared to the Chang et al. (2022) data set, which ended in 2020.

This part of the analysis places the focus on free tropospheric ozone variability and trends at 700–300 hPa over the period 1994–2021 in two domains (surface data are not included):

1. California: including THD ozonesondes, TMF lidar, AJAX flights, and IAGOS flights collected from San Francisco, Los Angeles, San Diego, San Jose, Fresno, and Las Vegas (this airport is geographically close

to California). Because we treat the AJAX and IAGOS flights as a single data set (see later discussion), this region consists of a total of three data sets;

2. WNA: includes the THD ozonesondes and TMF lidar data sets described above, plus the ozonesonde data from Boulder, Edmonton, and Kelowna/Port Hardy (these two sites are combined as a single data set), as well as other IAGOS data over WNA (e.g., Vancouver, Portland, Phoenix, Denver, and Seattle, combined with aircraft data from California as a single data set). This region consists of a total of six data sets.

Note that the study region over WNA is different from Chang et al. (2022), which did not include the AJAX data and IAGOS data collected over the Southwest U.S.. To be explicit, Chang et al. (2022) included the same ozonesonde and lidar data used in this study, but only incorporated the IAGOS data from San Francisco, Vancouver, Portland, and Seattle.

In order to better understand the difference in ozone trends between the free troposphere and the surface, several rural surface stations are included in the analysis. We use the hourly surface ozone data measured at the Mt Bachelor Observatory (MBO, 44.0°N, 121.7°W, 2,731 m), Oregon, which has been operated by the University of Washington (UW) since 2004. NOAA Global Monitoring Laboratory (GML) has operated a separate ozone instrument at the site. It should be noted that the GML instrument appears to have a positive bias in heavy smoke plumes (Bernays et al., 2022); for this reason, we use the UW ozone data, which have been processed and calibrated consistently through the data record (L. Zhang & Jaffe, 2017). Trend detection is based on daily time series aggregated by nighttime data only (8:00 p.m. to 7:59 a.m. local time or 4:00 a.m. to 15:59 p.m. UTC) to avoid upslope flow which transports ozone-depleted air from the surrounding valleys up to the summit. At night the localized mountain-valley circulations are diminished and MBO is more likely to sample baseline air from the Pacific Ocean, or polluted USA air that has been transported long distances (such as from California).

The analysis of rural boundary layer trends across the western USA relies on 26 high-elevation rural monitoring sites (see Table S1 in Supporting Information S1), maintained by the EPA Clean Air Status and Trends Network (CAST-NET) and the National Park Service. The daily observations were downloaded from the EPA data website (see data availability section) as maximum daily 8 hr average values (MDA8). This metric focuses on the time of day when the boundary layer is well mixed and therefore more likely to be regionally representative, and avoids the nighttime hours when the shallow nocturnal boundary layer leads to localized ozone deposition (Cooper et al., 2012). While all of these sites are located between 1.0 and 3.2 km above sea level, none are mountaintop sites. In contrast, MBO at 2.8 km above sea level is a mountaintop site, and as described earlier, the nighttime observations are made in laminar flow that lies above the nocturnal boundary layer and can, at times, be representative of the lower free troposphere or of baseline air that has been transported to the summit from the Pacific Ocean (Fischer et al., 2011; Gratz et al., 2015). However, the site is known to be heavily influenced by forest fires in summer and air masses at the site may have been influenced by the U.S. boundary layer during the previous day or days. Therefore, while we expect MBO to sample a different set of conditions and air masses from the other 26 rural surface sites, and while it should be more strongly influenced by the lower free troposphere, some influence from the U.S. boundary layer is still likely to occur. The links for all the data sets are provided in the data availability section.

2.2. Method

2.2.1. Data Fusion of Free Tropospheric Measurements From Multiple Platforms

Deriving the regional ozone trends in the free troposphere based on infrequent observations at sparsely distributed monitoring stations is challenging, due to multiple sources of uncertainty associated with limited data (Chang et al., 2022; Cooper et al., 2010; Lin et al., 2015). In our previous work, we developed a statistical framework for combining various non-independent vertical profile records, such as data sets from commercial aircraft, lidar, and ozonesonde measurements (Chang et al., 2022). The methodology is designed to, (a) determine the contribution of each data source through its deseasonalized and normalized deviations (ND), according to the sampling frequency and inherent data uncertainty; and (b) borrow the strength from neighboring (vertically) correlated structure to identify systematic ozone variability between data sets, remove unstructured variations within each data set, and thus achieve an enhancement of signal-to-noise ratio (Chang et al., 2020).

The rationale of data fusion developed in Chang et al. (2022) can be summarized by the following statistical structure:

various data sources = fused consensus process + discrepancy from individual dataset + random noise

(see Chang et al., 2022 for technical details) This methodology is implemented through the framework of the generalized additive models (Hastie & Tibshirani, 1990; Wood, 2006), by assigning separate penalized regression spline functions to the consensus process and discrepancy term. The regression splines are chosen so that the systematic variations between different data sets can be constituted (Wood et al., 2017). Therefore, this methodology is designed to separate common regional variations from each individual data source.

For additional implementation details, it should be noted that the methodology in Chang et al. (2022) was only applied to monthly mean deviations above WNA, but the variation in extreme percentiles is of particular interest to the research community, due to their relevance for impacts on human health and vegetation (Fleming et al., 2018; Mills et al., 2018). However, the statistical properties of extreme percentiles can be very different from the sample average (Berrocal et al., 2014; Eastoe & Tawn, 2009; Smith, 1989), thus the same methodology is not directly applicable. More specifically, Chang et al. (2022) used an inverse of standard error of the sample mean to represent the uncertainty associated with sample frequency and data variability, but this statistic is not necessarily representative of the uncertainty of the extreme percentiles (especially ozone data, which is often non-normally distributed), and might not be appropriate for weighting extreme percentiles from different sources of data sets. Even though the uncertainty of extreme percentiles can be assessed through bootstrap-based approaches (Gilleland, 2020; Kyselỳ, 2008), the weighting scheme for extreme values is currently absent and has not been thoroughly investigated. Therefore, in this study, we treat each data set equally when fusing the percentiles from different data sets.

While the equal-weight strategy for different data sets may seem to be less sophisticated, it is a desirable feature in this study. Recall that six data sets are used for data fusion, including Edmonton (53.6°N), Kelowna (49.5°N) combined with Port Hardy (50.4°N), THD (41.1°N), Boulder (40.0°N), TMF (34.4°N), and aircraft data combined from the IAGOS, AJAX and SNAAX (their sampling latitudes vary over time). Therefore 5 out of 6 data sets are sampled at fixed latitudes (accounting for 83% of the weights in the data fusion, the impact of the remaining 17% aircraft data on regional percentile trends will be discussed later). This weighting scheme preserves sufficient influence from each site spread over different latitudes. For example, the number of TMF lidar profiles in 2021 is roughly equal to the sum of the ozonesonde profiles added from Edmonton, Port Hardy, Boulder, and THD, so the equal-weight scheme in the algorithm, this does not imply each data set has the same contribution to the fused product, since the fused product will be determined from the common systematic variations between different data sets.

Spatial heterogeneity is an unavoidable complication in modeling regional ozone trends. The most straightforward approach is to explicitly account for the spatial variability (e.g., Chang et al., 2017, 2021). However, since the spatial coverages of free tropospheric observations vary greatly between different years above WNA (see Figure 1), aircraft data and observations at a few fixed locations are generally insufficient to determine the full spatial patterns. In terms of spatial representativeness, Liu et al. (2009) found the free tropospheric ozone observations are correlated over distances of 500–1,000 km. On the other hand, Stauffer et al. (2022) recently examined the stability of global ozonesonde network and concluded that those data are suitable to study trends (albeit the accuracy depends on the sampling frequency in each station), but trends at the individual station would much likely reflect localized variability (and partial regional variability). Therefore it is important to consider all available data sources as a whole for better characterizing large-scale variability (Chang et al., 2022; Cooper et al., 2010; Stauffer et al., 2022). Even though spatial heterogeneity is not fully accounted for in our current methodology, we evaluated the sensitivity of our data fusion method to spatial heterogeneity by also calculating the free-tropospheric ozone trends using a geographically weighted modeling approach. This second method is described in Section S1 in Supporting Information S1 and the results are discussed in Section 3.3, demonstrating that similar trends are produced by the data fusion and the geographically weighted methods.

2.2.2. Percentile Trend Estimations

As discussed above, conventional trend analyses of tropospheric ozone time series have been carried out using monthly, seasonal, or annual aggregated values (depending on the ozone metrics). The statistical purpose of such aggregations is to diminish heterogeneous data variability (e.g., hourly or daily values are expected to be even more variable and heterogeneous than monthly values), allowing traditional hypothesis tests to gain statistical power. On the other hand, in response to the need for understanding the trends of extreme events, previous attempts were often made by first deriving a single percentile per temporal scale, and then applying the same

6 of 26



1

trend model to the resulting ranked value time series. However, when we consider the above two issues together, the well-developed quantile regression method becomes a practical solution for both challenges. Quantile regression has been shown to be a well-suited approach for detecting heterogeneous distributional changes (Chang et al., 2021). The percentile trends can be quantified by directly applying the methodology to daily surface data or sparsely sampled ozone profiles, which is advantageous because no further data aggregation is needed, and we can keep as much information as possible for improved analysis of the extreme events (Gaudel et al., 2020).

The trend estimation is based on the following model, with adjustments for the potential correlation of the El Niño-Southern Oscillation (ENSO) and quasi-biennial oscillation (QBO) with free tropospheric ozone (Chang et al., 2022; Neu et al., 2014; Ziemke et al., 2019):

$$\psi_t = \beta_0 + \beta_1 t + \beta_2 \text{ENSO} + \beta_3 \text{QBO}_{30hPa} + \beta_4 \text{QBO}_{50hPa} + \varepsilon_t, \tag{1}$$

where y_t is the ozone time series with a temporal index t, β_0 is an intercept, β_1 is the linear trend estimate, $\{\beta_2, \beta_3, \beta_4\}$ is a set of coefficients associated with ENSO and QBO at 30/50 hPa, respectively, and ε_t is the random noise (assumed to be an AR(1) process). Note that the data are deseasonalized in advance, so the seasonal component is not included in Equation 1.

Equation 1 describes how the statistical relationship is formulated, the next step is to provide context for evaluation of the trend and regression coefficients by median- and quantile-based methods. Median regression (also known as least absolute deviations, LAD) is well known for its robustness to outliers and non-normally distributed data, as compared to the least squares (mean-based) method. Quantile regression extends LAD by enabling quantification of other percentile changes (Koenker & Hallock, 2001). If we rewrite Equation 1 as $\varepsilon_t = y_t - X\beta$, where t = 1, ..., T, then the percentile estimate for the trend and each regression coefficient can be evaluated by minimizing the following weighted least absolute error criterion:

$$\sum_{\{t|\epsilon_t \ge 0\}}^{T} q|y_t - X_t \beta| + \sum_{\{t|\epsilon_t < 0\}}^{T} (1-q)|y_t - X_t \beta|,$$
(2)

where 0 < q < 1 represents the quantile to be estimated (e.g., q = 0.5 represents the median).

Although autocorrelation tends to cause an underestimation of uncertainty associated with the trend estimate (Weatherhead et al., 1998), the influence of daily autocorrelation on the long-term trends is expected to be rather weak. On the other hand, bootstrapping methods are known to produce a consistently greater estimation uncertainty than standard methods, when data are heteroscedastic and/or non-independent (Gonçalves & White, 2005; Politis et al., 1997). Therefore, in order to account for as much variability as possible, we adopt a bootstrapping approach to determine the standard error associated with the trend (i.e., uncertainty estimate) from quantile regression (Koenker, 1994). In this study, we use SNR (signal-to-noise ratio, defined as the ratio between trend and its uncertainty) value to assess the reliability of the trend estimate. An SNR value of 2 (or 3) represents the threshold of approximately the 95% (or 99%) confidence level. The higher the absolute SNR value, the higher the confidence of the trend estimate.

It should be emphasized that in this study quantile regression is only applied to surface measurements at each monitoring site. For the free tropospheric ozone observations from different platforms, we first account for the heterogeneity between different data sets. Therefore, the data fusion described in Section 2.2.1 is applied to the profile data from each data set in order to produce the fused ozone percentile (vertical) distribution over time, and then the mean trend estimates are derived from the resulting fused percentile product. To be consistent with the uncertainty from quantile regression, the uncertainty of the trend estimate for the fused percentile time series is determined by the bootstrapping method implemented by the R package boot (Canty, 2002; Kunsch, 1989).

3. Results: Trend Detection of Free Tropospheric Ozone Based on the Data Fusion of Ozonesonde, Lidar, IAGOS, and AJAX Data Sets

3.1. Preliminary Analysis

To ensure the field campaign data can be incorporated into the trend assessment, we evaluated the comparability between different data sets to avoid suspicious jumps or offsets, which can bias the trends. Field campaign data are usually scheduled for certain years, which makes it difficult to derive reliable long-term monthly means to

deseasonalize the data. Since our methodology fuses data through the deseasonalized and ND, we assume the AJAX data (in the free troposphere) can be deseasonalized by the seasonal cycle based on the IAGOS data. Figure S1 in Supporting Information S1 shows the boxplots of annual ozone observations and their deseasonalized anomalies from different data sets in California over 2011–2020. We can see some discrepancies between the annual distributions of the AJAX (as well as IAGOS) and ozonesonde/lidar data (e.g., 2014 and 2017), mainly because the AJAX data are not uniformly sampled for all months, but after deseasonalization, the discrepancies are diminished. This result indicates that the AJAX data are compatible with other data sets and can be incorporated into the trend assessment. Despite the fact that the IAGOS data set has a large gap over 2006–2016 in California, incorporation of the AJAX data set partially fills the data gap and produces a more accurate trend estimate.

Trend analysis without taking seasonality into account (or deseasonalization) is not only prone to estimation bias (when data coverage is limited in some years), but also much more likely to produce an inflated trend uncertainty (regular patterns should not be considered to be a part of trend uncertainty). Figure S2 in Supporting Information S1 shows the median ozone time series from each data set, with a linear trend estimate and a Loess (locally estimated scatterplot smoothing, Cleveland, 1979) nonlinear fit to the deseasonalized anomalies. Even though different linear trends and uncertainties are found for different data sets (due to localized differences), the general variability is consistent among time series, except for the early period where the confidence intervals between IAGOS and ozonesonde data are separated for a short period of time (but this does not adversely affect the trend assessment, as indicated by a sensitivity test of trends derived from different starting years; see Section 3.3). Indeed, even though IAGOS can be considered to be a reference data set due to its known calibration history (Tarasick et al., 2019), and ozonesonde data are of high quality and stability (Stauffer et al., 2022), Tarasick et al. (2019) concluded that ozonesonde observations are moderately higher (5%–8%) compared to IAGOS data. This mean difference could produce a bias in the trend estimate, but deseasonalization of each data set removes this offset.

The merged annual ozone data set and its anomaly distributions based on all available data over California are shown in Figure 2. We can see how the ozone distribution varies across the years, and this plot also demonstrates the importance of deseasonalization for time series analysis, as it transforms non-normal distributions (presumably due to a non-uniform or sporadic sampling schemes) into normal-alike distributions (see the distribution in 2003 and 2004 as an example). However, a positive skewness can still be observed in some deseasonalized anomaly distributions; in this case, a difference between mean and median trends might be expected (i.e., heteroscedasticity), and an investigation of the extreme percentiles is desirable for better understanding of a greater extent of the ozone variability.

3.2. Mean Trends

Since average ozone mixing ratios and their variations typically increase with altitude, we transform the ozone profile data into ND by deseasonalizing and normalizing ozone measurements at each 10 hPa pressure surface, so that the data are more comparable across different pressure levels and different data sets. This consideration is critical because statistical models perform better when data have similar levels of variability. The data fusion methodology is then applied to the deviations from each data set.

Figure 3 shows the fused ozone distributions in the free troposphere, based on the ND and a transformation back to the units of ppbv, over California and WNA respectively. The plot of ND enables us to enhance the delicate ozone variability across different pressure surfaces; we can observe that the general patterns are well correlated between California and WNA in terms of the distribution of interannual positive and negative anomalies. To avoid the impact of incomplete data in 1994 above California and anomalously low ozone in 2020 (Chang et al., 2022; Steinbrecht et al., 2021), the trend estimates are reported for three periods 1995–2019, 1995–2020, and 1995–2021, and based on the column average over 700–300 hPa of the fused monthly anomalies (e.g., the last two panels of Figure 3). The positive trends above California are found to be slightly stronger than WNA, but the interannual variability is similar in general between the two regions. The California data are a subset (50%) of the WNA data set, and cover a much smaller region. The fact that similar interannual variability and SNR (signal-to-noise ratios) can be derived from the California sub-domain indicates that the observed ozone increase is a robust feature of the WNA domain (see Table 1).

To highlight the fact that intermittent field campaign data available in certain years can still add value to regional trend detection, Figure S4 in Supporting Information S1 shows the fused ozone distributions above California, with and without the AJAX data. With additional data included in the analysis, we can see a much more detailed variability in 2011–2018 with an enhanced SNR value in trend detection.





Annual ozone densities over time above California, 700-300 hPa

Figure 2. Annual ozone densities above California (1995–2021), from all available observations (left panel) and deseasonalized anomalies (right panel). Data are deseasonalized by harmonic functions fitted from each data source (THD sondes, TMF lidar, and aircraft data combined from IAGOS, AJAX, and SNAAX).

To compare with our previous result (Chang et al., 2022), we provide the trend estimates with reference to 1994 in Table S2 in Supporting Information S1. Our previous regional mean trend estimate of free tropospheric ozone over WNA was $0.4 [\pm 0.2]$ ppbv/decade (1994–2019), but as mentioned in Section 2.1, our previous result did not include the AJAX data or the IAGOS data over the Southwest (i.e., data collected from Las Vegas, Los Angeles, and San Diego). The regional mean trend above WNA in this study is much stronger (1.0 [\pm 0.3] ppbv/decade) over the same period. We conclude that the discrepancy with our previous trend estimate is due to a greater ozone enhancement in the second half of the study period revealed by additional data above the Southwest.

The following analysis places the focus on the comparison of a specific portion of ozone variability (i.e., percentile and seasonal trends), but we should keep in mind that the overall trends remain fairly consistent between California and WNA.

3.3. Percentile Trends

The transport and chemical processes that drive heterogeneous trends are typically different for high and low ozone in the free troposphere. The high ozone values are more likely affected by stratosphere-troposphere exchange processes (Langford, 1999), long-range transport of air pollution plumes (Nowak et al., 2004), or wildfires (Jaffe & Zhang, 2017), whereas the lowest ozone values above WNA typically originate in the tropics (Asman et al., 2003;





Figure 3. Fused ozone mean (anomaly) distributions in the free troposphere above California and western North America, based on normalized deviations (ND) and in units of ppbv.

Davies et al., 1998; Grant et al., 2000). The data fusion methodology is applied to the 5th, 50th, and 95th percentiles, and the resulting fused distributions above California are provided in Figure 4: the 95th percentile trend is positive and stronger than the 50th percentile, and then followed by the 5th percentile. To better compare the variability from different percentiles, the curtain plots based on units of ppbv are scaled to their relative percentile values (instead of zero centered anomalies as in Figure 3). Noticeable common interannual variabilities include.

• The enhanced ozone episode over 1997–1998, previously attributed to enhanced stratosphere-troposphere exchange following a strong El Niño event (Koumoutsaris et al., 2008; Langford, 1999), but also explained by extreme biomass burning in southeast Asia (Fiore et al., 2022);



Table 1

Ozone Trends and Uncertainties [in Units of ppbv/decade] Based on the Column Average Over 700–300 hPa of the Fused Monthly Anomalies (With the Vertical Correlations, Data Uncertainty and Variability Accounted for) Above California and Western North America (WNA), and Rural Surface Sites Across WNA

		1995–2019		1995–2020		1995–2021	
		Trend $[\pm 2 - \sigma]$	SNR	Trend $[\pm 2 - \sigma]$	SNR	Trend $[\pm 2 - \sigma]$	SNI
California (free troposphere)	Mean	1.2 [±0.4]	7.1	0.9 [±0.4]	4.7	1.0 [±0.3]	5.0
	95th	2.1 [±0.6]	6.9	2.1 [±0.6]	7.3	2.6 [±0.6]	8.
	50th	1.8 [±0.3]	10.9	$1.4 [\pm 0.4]$	7.0	1.4 [±0.3]	8.0
	5th	1.7 [±0.4]	8.0	0.9 [±0.6]	3.0	0.5 [±0.5]	2.0
WNA (free troposphere)	Mean	1.0 [±0.3]	7.2	0.7 [±0.3]	4.7	0.7 [±0.3]	4.8
	95th	0.4 [±0.5]	1.5	0.2 [±0.5]	1.0	0.5 [±0.5]	1.8
	50th	0.7 [±0.3]	5.4	0.5 [±0.3]	3.4	0.5 [±0.2]	4.3
	5th	0.6 [±0.3]	4.4	0.2 [±0.3]	1.0	0.1 [±0.3]	0.4
WNA (rural surface)	Mean	$-0.9 [\pm 0.4]$	-5.2	$-1.0 [\pm 0.4]$	-5.8	$-1.0 [\pm 0.3]$	-6.0
	95th	-1.4 [±0.5]	-5.9	-1.4 [±0.4]	-6.5	$-1.4 [\pm 0.4]$	-6.9
	50th	$-0.9 [\pm 0.3]$	-5.1	$-1.0 [\pm 0.4]$	-5.6	$-1.0 [\pm 0.3]$	-5.9
	5th	$-0.6 [\pm 0.3]$	-3.2	$-0.7 [\pm 0.3]$	-4.0	-0.6 [±0.3]	-3.9
California (free troposphere)	DJF	1.9 [±0.6]	6.2	1.7 [±0.6]	5.7	1.5 [±0.6]	5.0
	MAM	2.2 [±0.7]	6.0	2.0 [±0.7]	5.5	1.6 [±0.8]	4.2
	JJA	1.8 [±0.8]	4.4	1.6 [±0.8]	4.1	1.9 [±0.8]	4.9
	SON	0.9 [±0.3]	5.1	0.8 [±0.3]	4.9	0.9 [±0.3]	5.4
WNA (free troposphere)	DJF	0.9 [±0.5]	4.0	0.8 [±0.4]	3.7	0.7 [±0.4]	3.2
	MAM	0.8 [±0.6]	2.8	0.6 [±0.6]	2.0	0.4 [±0.6]	1.5
	JJA	0.4 [±0.5]	1.8	0.3 [±0.5]	1.2	0.5 [±0.5]	2.2
	SON	$-0.7 [\pm 0.6]$	-2.4	$-0.8 [\pm 0.5]$	-3.1	-0.7 [±0.5]	-2.7
WNA (rural surface)	DJF	0.02 [±0.3]	0.1	$-0.02 [\pm 0.3]$	-0.1	-0.1 [±0.3]	-0.0
	MAM	$-0.5 [\pm 0.6]$	-1.8	$-0.9 [\pm 0.6]$	-2.8	-0.9 [±0.6]	-3.2
	JJA	$-2.0 [\pm 0.8]$	-5.3	$-2.2 [\pm 0.7]$	-6.0	-1.9 [±0.7]	-5.2
	SON	$-1.0[\pm 0.4]$	-5.8	-0.9 [±0.3]	-5.4	-0.9 [±0.3]	-5.9

Note. SNR (signal-to-noise ratio) is the ratio between trend and its standard error.

- An enhancement in 2015 in the 50th and 5th percentiles. An attribution study of this free tropospheric enhancement has not yet been conducted, however, ozone enhancements at the surface and in the boundary layer of the western USA have been linked to a high-temperature anomaly and increased biogenic emissions (Jaffe & Zhang, 2017);
- Strong ozone anomalies in 2018 in the 95th, 50th and 5th percentiles. An attribution study of this free tropospheric enhancement has not yet been conducted. A previous study found strong correlation between enhanced ozone and fires for 2017, 2018, and 2020 in urban areas over WNA (Jaffe et al., 2022), and we suggest that a future attribution study could explore the possibility that wildfire smoke impacted free tropospheric ozone.

The corresponding fused distribution over WNA (which includes the California subdomain) is shown in Figure 5: the magnitude of the trend in all three percentiles is weaker with respect to California, and the 5th percentile trend is nearly flat over 1995–2021, but the interannual variability shows good agreement with the California subdomain.

Figure S5 in Supporting Information S1 shows the changes of trends presented by 10 hPa resolution vertical profiles for both regions based on the ending year of 2019, 2020, and 2021, respectively. The vertical structures of the percentile trends do not reveal any obvious differences due to the choice of ending year. Therefore, we summarize the overall free tropospheric ozone trends (700–300 hPa) in all three percentiles in the first part of





Journal of Geophysical Research: Atmospheres



Figure 4. Fused ozone percentile (anomaly) distributions in the free troposphere above California, based on normalized deviations and in units of ppbv. Trend values are referenced to 1995–2021.

Table 1. We can observe a coherent drop of trends for both regions when 2020 data are included, except for the 95th percentile above California (which is presumably associated with extensive wildfires in 2020 as discussed above). When the study period is extended to 2021, we can observe increasing positive trends in the 95th percentiles, no change in the median trends, and much weaker positive trends in the 5th percentile.

To investigate the heterogeneity of regional percentile trends, we further extend the analysis by applying the same statistical methodology to every 5-percentile interval between the 5th and 95th percentiles. The resulting distributions of percentile trends are shown in Figure 6. The corresponding curtain plots of ozone distributions by every 10th percentile are shown in Figure S6 in Supporting Information S1 (above California) and S7 (above WNA). We summarize the findings as follows:

- At first glance, we can see that neither California nor WNA shows a strictly monotonic or homogeneous structure of percentile trend distributions, which might indicate that the 5th, 50th, and 95th percentiles may not be sufficient to fully represent the heterogeneous ozone variability. For example, the strongest positive trends over WNA are found within the 10th–20th percentile range, but the 5th percentile trends are nearly flat over 1995–2021.
- Prior to the COVID-19 economic downturn (1995–2019), nearly all percentiles above California and WNA were reliably positive. Inclusion of the pandemic years (1995–2021) shows that the positive trends in the lower percentiles (5th–25th percentiles) above California were greatly diminished.
- Above California, all trends between the 95th and 5th percentiles remain reliably positive (SNR > 2, i.e., the lower bound of the 2- σ interval greater than zero) after the 2020 COVID-19 impact. Above WNA, despite the reduction in the lower percentile trends, the general pattern of percentile trend distribution has not changed (e.g., trends in the 10th–70th percentile range remain reliably positive.





Figure 5. Fused ozone percentile (anomaly) distributions in the free troposphere above western North America, based on normalized deviations and in units of ppbv. Trend values are referenced to 1995–2021.

Figure 7 summarizes the overall quantified annual percentile anomalies in the free troposphere, and the merged monthly median time series (with observed fluctuations), above California and WNA, respectively. Note that the magnitudes of the 95th and 5th percentiles are average differences from the medians. The annual median anomalies are strongly correlated (r = 0.84) between California and WNA, followed by the 5th (r = 0.72) and 95th



Figure 6. Changes of percentile trend distributions in the free troposphere above California and western North America. Dashed lines represent the $2-\sigma$ uncertainty.





Figure 7. Quantified ozone percentile anomalies and merged median time series measurements in the free troposphere above California (CA) and western North America (WNA). For the anomaly plots, (a) uncertainty represents the 2-sigma range; and (b) the magnitudes of the 95th and 5th percentiles are average differences from the medians.

(r = 0.66) percentiles. Note that the uncertainty for the 95th percentile is constantly greater than the 50th and 5th percentiles, this is not unexpected due to a positive skewness and a heavy tail of the ozone distribution.

As pointed out previously, ozonesonde and lidar data have a fixed weight in the data fusion (accounting for 83% of the regional contribution), therefore a 17% weight is contributed by aircraft data sampled at varying latitudes. Figure S8 in Supporting Information S1 shows the annual data density made by the sampling latitudes from the IAGOS and AJAX data. Since 2016 there are fewer data sampled from high latitudes (over >45°N), which could potentially create a bias by being heavily weighted toward low latitude data. Nevertheless, the downward tendency of the 5th percentile anomalies over 2015–2020 can be clearly observed above California (the top panel of Figure 7), but not observed over a greater domain above WNA, this indicates the regional trends over WNA should not be heavily impacted by imbalanced sampling latitudes from aircraft data in recent years.

To examine the reliability of our methodology for deriving fused percentile trends, in Section S1 in Supporting Information S1 we adopt a geographically weighted modeling approach to derive the overall trends by an adjustment to the spatial heterogeneity, and in Section S2 in Supporting Information S1 we apply different approaches for estimating the regional ozone percentiles (Fasiolo et al., 2020; Hyndman & Fan, 1996). Both results show great similarity to the discussion above, confirming the effectiveness of our approach. Results from a further sensitivity analysis are presented in Figure S12 in Supporting Information S1 to investigate the percentile trend estimates and uncertainties based on different starting years (1995, 1998, 2000, and 2005) and ending years (2019 and 2021). The year 1998 is selected as it represents a well-known enhanced ozone episode that was observed



across the Northern Hemisphere, and has been recently attributed to extreme biomass burning (Fiore et al., 2022). In general, the big picture remains similar, except that, (a) the lower percentile trends above California are shifted from positive over 1995–2021 toward negative over 2005–2021 (as discussed above, lower percentile trends in California are heavily affected by the pandemic years); and (b) if the estimates are based on the starting year in 1998 with unusually high ozone, its impact can be observed mainly on the reduction of trends at the higher percentiles above WNA, but the sensitivity is weak at the lower percentiles. Nevertheless, with the exception of lower percentile trends above California, the trends and uncertainties are comparable regardless of the starting year (1995, 2000, or 2005), and we conclude that our result is robust.

3.4. Seasonal Trends

Figure 8 provides seasonal ozone distributions and trends (1995–2021) based on the fused medians from different data sets above California and WNA (the corresponding curtain plots for ND are provided in Figure S13 in Supporting Information S1). In California, all seasons show positive trends (very high confidence, SNR > 4), with the strongest trend observed in JJA (June–July–August), followed by MAM (March–April–May), DJF (December–January–February), and SON (September–October–November). In WNA, we find strong positive trends for DJF and JJA (high confidence, SNR > 2), weak positive trends for MAM, and strong negative trends for SON. In both regions, the most confident (i.e., the highest SNR value) positive trends occur in DJF. The magnitude and SNR of the trends above WNA are weaker than California.

A notable feature in Figure S13 in Supporting Information S1 and Figure 8 is that the interannual variability is more dynamic in MAM and DJF. To explore the reason behind this phenomenon, additional analyses are provided as follows:

- We investigate the stratospheric influence on tropospheric ozone by showing the percentage (from all available profiles) of ozone values exceeding a threshold at 100, 125, 150, 175, 200, and 225 ppbv above WNA, respectively in Figure S14 in Supporting Information S1; such high ozone values in the free troposphere are explained by stratospheric intrusions. The frequency of observing ozone exceedance across 700–500 hPa seems to be more common in recent years than the late 1990s. We made similar plots focusing on MAM and JJA in Figure S15 in Supporting Information S1: For a threshold of 150 ppbv and in the 350–300 hPa layer, 42 months (50%) in JJA and 10 months (12%) in MAM have zero exceedances over 1995–2021. For the months with non-zero exceedance rates, an averaged exceedance rate in MAM is also higher than JJA. The greater frequency of stratospheric intrusions in spring compared to summer is consistent with many previous studies (Cooper et al., 2014; Stohl et al., 2003). Given that these high ozone values occur in very few profiles in any given month, any impact on ozone trends would be limited to the higher percentiles.
- We extracted meteorological output at San Diego, THD, and Vancouver from the NASA Global Modeling and Assimilation Office's MERRA-2 (Modern-Era Retrospective analysis for Research and Applications, Version 2) reanalysis (Gelaro et al., 2017), including seasonal geopotential height and specific humidity at 500 hPa (Figure S16 in Supporting Information S1) and monthly mean and standard deviation of potential vorticity (PV) at 300 hPa (Figure S17 in Supporting Information S1). The variability of PV at all three locations is highest in DJF and MAM, indicating a greater frequency of upper-level long wave troughs at this time of year, which are commonly associated with stratospheric intrusions and therefore high ozone variability, consistent with the extreme ozone observations described above. Geopotential height and specific humidity are greatest during summer at all three locations, indicating a higher frequency of upper-level ridges and moist tropical air extending into mid-latitudes. These conditions are consistent with a northward migration of long-wave troughs to high latitudes, minimizing the occurrence of stratospheric intrusions. These dynamical indicators point to conditions in summer that are relatively quiescent with increased tropical characteristics, which limit ozone variability.

The lower half of Table 1 reports the seasonal trend estimates for both regions. The seasonal positive trends over 1995–2019 diminish after the 2020 data are included, but overall, the impact of the 2020 COVID-19 economic downturn on tropospheric ozone trends is slight in SON above California and more pronounced in other seasons. When the 2021 data are included, the diminished trends persist in DJF and MAM, but there was a rebound of the positive trend in JJA (consistent with the findings from three NASA satellite products (Ziemke et al., 2022)). Above WNA Cooper et al. (2010) found the median trend of free tropospheric ozone in springtime to be 0.63 $[\pm 0.34]$ ppbv/season over 1995–2008. Lin et al. (2015) updated the analysis to 1995–2014 and found a weaker





Figure 8. Fused ozone (anomaly) distributions of seasonal medians in the free troposphere above California and western North America, based on deseasonalized anomalies (in units of ppby). Trend values are referenced to 1995–2021.

trend of 0.31 [\pm 0.21] ppbv/season. Our updated result shows a similar trend of 0.33 [\pm 0.25] ppbv/season (or 0.08 [\pm 0.06] ppbv/yr) over 1995–2019 and a weaker trend of 0.17 [\pm 0.22] ppbv/season (or 0.04 [\pm 0.06] ppbv/yr) over 1995–2021.

4. Result: Rural Surface Ozone Trends

The previous section focused on a data fusion of the free tropospheric ozone profiles from several different data sets, with trend estimates derived from the fused product. This section will focus on surface ozone observations and trends.





Figure 9. Ozone variability and trends at Mt Bachelor based on nighttime measurements: (top left) mean and percentile seasonality based on UW observations over 2004–2020; (top right) changes of percentile trends over different periods; (middle) percentile trends derived by mean regression applied to percentile time series; and (bottom) percentile trends derived by quantile regression applied to all data points.

4.1. Surface Trends at the Mt Bachelor High Elevation Site

As described above, heterogeneous ozone trends and variability at different percentiles were found in the free troposphere in association with the different data platforms, which had to be taken into account when applying the data fusion technique. However, if the time series data are from a single source (e.g., the same location and same measurement technique), quantile regression is a straightforward and preferable approach for trend detection of ozone observations. Even though the applications of quantile regression have gained popularity in the trend detection of atmospheric composition and pollutants (Baur et al., 2004; Gaudel et al., 2020; Munir et al., 2012; Sousa et al., 2009; Xu & Lin, 2018), the practical adjustments for atmospheric time series analysis, such as the potential impact of inconsistent seasonality at different percentiles on the percentile trend estimate, have not been investigated. Therefore, we provide a comprehensive technical evaluation of quantile regression specifications in Supporting Information S1 as follows:

Section S3 Supporting Information S1 demonstrates that deseasonalization is a critical consideration for accurate percentile trend detections. As a demonstration of quantile regression, Figure 9 shows the (deseasonalized) percentile time series at MBO with their mean trend estimates, as well as daily time series with percentile trend estimates derived by quantile regression. We can see that these two approaches produce a consistent



Table 2

Ozone Trends [in Units of ppbv/decade] Based on the Quantile Regression at Mt Bachelor Observatory Since 2004

	2004–2019		2004–202	0	2004–2021		
	Trend $[\pm 2 - \sigma]$	SNR	Trend $[\pm 2 - \sigma]$	SNR	Trend $[\pm 2 - \sigma]$	SNR	
Mean	3.2 [±0.6]	10.4	1.6 [±0.6]	5.7	2.4 [±0.5]	9.0	
95th	3.5 [±1.7]	4.2	2.6 [±1.6]	3.2	4.6 [±1.5]	6.4	
50th	2.7 [±0.6]	8.9	1.7 [±0.6]	5.8	2.0 [±0.6]	6.6	
5th	4.0 [±1.7]	4.6	1.4 [±1.4]	1.9	1.9 [±1.4]	2.7	

Note. SNR (signal-to-noise ratio) is the ratio between the trend and its standard error.

median trend estimate, but some discrepancies are found at the 5th and 95hr percentile trends, likely due to a short record being considered here (18 yr). These discrepancies are expected to be reconciled when long-term data are available.

A series of sensitivity tests is presented in Section S4 in Supporting Information S1 to demonstrate that (moderate) intermittent data gaps are expected to have a small impact on the percentile trend estimates. Since the above inconsistency in the extreme percentile trends is very likely due to a short data record, and since quantile regression is tailored to the purpose of detecting distributional changes (Reich, 2012), we therefore adopt quantile regression to produce more robust percentile estimates. Section S5 in Supporting Information S1 provides the details on how an attribution of data variability can be performed at different percentiles.

To investigate the impact of the 2020 COVID-19 economic downturn on ozone trends at MBO, Table 2 provides the trend estimate, $2-\sigma$ uncertainty and SNR value for the 5th, 50th, and 95 percentiles based on three periods, 2004–2019, 2004–2020, and 2004–2021, respectively. The positive trends over 2004–2019 are clearly weakened when 2020 data are included, with a 50% reduction in mean and a 37% reduction in median. We also see that the weakened trends rebounded in 2021, especially for the 95th percentile. Figure 10 shows the percentile trends over the above three periods, from the 5th to 95th percentile by every 5 percentile interval. In addition, it is well known that wildfires have an impact on ozone variability at MBO in summer (Baylon et al., 2015; Laing et al., 2016), thus we also show the same analysis, but exclude the wildfire season (June–September). The 2020 COVID-19 impact on percentile trends can be clearly observed in both scenarios: a stronger reduction is found in the lower percentiles and a weaker change is found in the higher percentiles (consistent with the free tropospheric results above WNA and California). Another noteworthy feature is that the positive trends in the 95th and 90th percentiles became much weaker over 2004–2021 when the data in the wildfire season were removed, indicating that the rebounding 95th percentile trend is very likely driven by ozone production associated with the extensive fires in 2021.

4.2. Surface Trends at Rural Sites Across the Western USA

Twenty-six high-elevation, rural monitoring sites across the western United States were selected to represent long-term, regional-scale ozone variability in the boundary layer; see Table S1 in Supporting Information S1 for site details (station name, state, longitude, latitude, and elevation). Some of these sites were previously analyzed by Cooper et al. (2012), who found increasing median ozone trends in MAM, and a range of trends in JJA for the period 1990–2010.









Rural surface ozone across western North America

Figure 11. Regionally weighted rural surface measurements in western North America, for 2020 (orange) and 2021 (purple) compared with individual years from 1995 to 2019 (blue) and the average over 2000–2019 (black). Monthly values are based on all available MDA8 ozone values.

Percentile trends for each site are shown in Figure S24 in Supporting Information S1 over 1995–2021 (any sites with measurements beginning after 2005 are excluded from this particular plot, so a total of 21 stations are shown). The trend values and their associated uncertainties are based on the 2000–2019 reference period and quantile regression of MDA8 ozone. The results show that 12 (57%), 17 (81%), and 20 (95%) sites have reliably strong decreasing trends in the 5th, 50th, and 95th percentiles, respectively, so that an overall decrease of regional ozone is not unexpected from this consistent pattern of negative trends across WNA.

To better summarize the regional variations across the domain, we applied a geographically weighted regression model to all available time series data, based on a Gaussian spatial process model fitted through the generalized additive models (Chang et al., 2017, 2021). The fitted result can be considered to be the overall common trend after an adjustment of the irregularity of the spatial distribution of monitoring stations. The resulting regional 5th, 50th, and 95th percentiles displayed by individual years from 1995 to 2021 are shown in Figure 11 (their continuous time series plots are shown in Figure S25 in Supporting Information S1). A clear ozone reduction was observed in MAM and JJA of 2020 in all of these percentiles, except for a strong enhancement in the 95th percentile that coincides with the wildfires in August 2020 (Filonchyk et al., 2022). A clear rebound of ozone in 2021 to 2000–2019 climatological mean can also be observed (except for MAM where the ozone level is still lower than the mean over 2000–2019). Over 1995–2021 the 95th percentiles have the strongest negative trends, followed by the 50th and 5th percentiles (Figure S25 in Supporting Information S1).

The percentile and seasonal trend estimates for the regional rural surface sites are provided in Table 1. Despite a clear ozone reduction observed in 2020, the negative percentile trends over 1995–2019 show little or no change when the 2020 and 2021 data are included. A greater impact of the 2020 COVID-19 economic downturn on trends is observed in MAM (from $-0.5 \ [\pm 0.6]$ ppbv/decade over 1995–2019 to $-0.9 \ [\pm 0.6]$ ppbv/decade over 1995–2021). When comparing the seasonal trends before and after 2020, weak changes in JJA/SON may reflect the transition from emissions reductions during the lockdown to ozone enhancement due to the wildfires in the late summer of 2020 (such a reflection within 2020 can also be observed in Figure 11).

4.3. Summary of Differences in Trends Between the Free Troposphere and Rural Surface Measurements

A highly consistent pattern in Table 1 is that all of the percentile trends show a substantial difference between the free troposphere and rural surface when considering all months in the year, that is, opposite in magnitudes and no overlap of the $2-\sigma$ uncertainties. For seasonal trends, the largest difference in trends is found in JJA, followed by MAM, and similar trends are only found in SON. This finding adds to the growing body of evidence that surface trends are frequently disconnected from the general increases observed in the free troposphere (Cooper et al., 2020), as assessed by IPCC (Gulev et al., 2021), and also reproduced by a recent chemistry-climate model simulation (Fiore et al., 2022).





Figure 12. Comparison of ozone median trends by month from the free troposphere above California and western North America (WNA), regionally weighted rural surface measurements in WNA, and Mt Bachelor data. Trend values are referenced to 1995–2021 and 2004–2021.

In order to summarize the overall difference in (median) trends between the free troposphere and the rural surface over WNA, we compare the trends at the rural surface sites to the free tropospheric trends above California and WNA, based on individual months for the periods 1995–2021 and 2004–2021 (with the MBO record included) in Figure 12. From the top panel (1995–2021) we can clearly observe opposite trends in summertime and inseparable trends in wintertime between the free troposphere and rural surface, which clearly indicates that different processes must be affecting the decreasing ozone in the WNA boundary layer compared to the increasing ozone in the free troposphere. When we limit the study period to a shorter time frame (2004–2021), similar seasonal patterns can still be observed. Consistent positive trends can be observed at MBO, with much stronger enhancements in August–October presumably due to the wildfire impacts (Baylon et al., 2015; Jaffe et al., 2022; Laing et al., 2016). A better agreement of trends between MBO and the free troposphere above WNA/California is observed in April–July.

5. Discussion

This study provides an extensive analysis of tropospheric ozone trends and variability over WNA, using (a) free tropospheric ozone profiles from ozonesondes, TMF lidar, IAGOS commercial aircraft, and the NASA AJAX field project; and (b) surface measurements from MBO, and the National Park Service and EPA CASTNET monitoring networks. Although we present the results starting with the free troposphere and then proceed to the rural surface sites across WNA, the trends are summarized for three different scenarios:

- 1. A surface time series measured at a fixed location: Since surface station data are usually measured at a high frequency (i.e., hourly values are available, albeit subject to data gaps), applying quantile regression to daily aggregated values is appropriate for studying the extreme events.
- 2. *Multiple surface time series collected from a regional monitoring network*: For each predetermined percentile, the spatial heterogeneity (from this specific percentile variability) and the irregular distribution of monitoring



locations need to be taken into account to produce the common regional time series, and then the overall trends can be derived.

3. *Multiple time series made by sparsely sampled vertical profiles from different platforms*: For each predetermined percentile, various data sets are combined (via a data fusion methodology) by taking into account the vertical correlation structures and data characteristics (from this specific percentile variability). The fused result is produced as the ozone vertical distribution over time, and the overall trends are derived from the column averages of the fused product.

All the percentile trends examined in this study show heterogeneity (i.e., consistent trends are not found between different percentiles and seasons). This result suggests that only using the mean and a few percentiles (e.g., 5th and 95th) may be insufficient to fully represent the heterogeneous ozone variability, given the fact that the distribution of percentile trends is not monotonic between the 5th and 95th percentiles above WNA. As a candidate for median trend estimators, quantile regression has the same advantage as the classical Sen-Theil estimator, such as robustness to outliers and no assumption on error distribution. As a part of regression-based methods, quantile regression has more flexible model structures over the classical non-parametric methods, such as incorporation of covariates for trend attribution. We show that percentile trend estimates based on quantile regression are resistant to moderate data gaps (even with a short data record), and consistent with the result estimated through individual ranked values (if data samples are sufficient). Therefore, quantile regression is considered to be the preferred approach, especially when data samples are sparse.

The trend analysis of the fused free tropospheric observations and rural surface network in this study can be summarized as follows:

- Despite being impacted by the 2020 COVID-19 economic downturn, median ozone trends remained positive in the free troposphere over 1995–2021. The interannual variabilities of median ozone above California and WNA are well correlated, but the magnitude of trends is much stronger above California (1.4 [±0.3] ppbv/ decade) than WNA (0.5 [±0.2] ppbv/decade).
- 2. In the free troposphere above California, nearly all percentile and seasonal trends are stronger and more confident than above WNA. The 2020 COVID-19 impact can be clearly observed from weakening trends in the lower and middle percentiles above California. After the 2021 data are included, we found that the positive 5th percentile was further diminished, with a rebounding positive trend in the 95th percentile, and no change in the median trend. The percentile and seasonal trends remain consistently positive (high confidence, SNR > 2) since 1995 through the end of 2021.
- 3. Weak positive mean and median trends can be detected reliably in the free troposphere above WNA over 1995–2021. While positive trends were also observed in the 5th and 95th percentiles, the confidence level is much lower; positive trends can also be found consistently (high confidence) between the 10th–70th percentile range. Even though weakened positive trends in the lower percentiles are found after the 2020 COVID-19 pandemic, the overall distribution of percentile trends remains similar.
- 4. After the 2020 COVID-19 pandemic, continuously weakened positive trends are found in DJF and MAM, and rebounding positive trends are found in JJA through 2021 in the free troposphere above WNA (as well as California). Since 1995, increasing ozone (high confidence) can still be observed in DJF (0.7 [±0.4] ppbv/decade) and JJA (0.52 [±0.47] ppbv/decade); trends in MAM have diminished from (0.8 [±0.6] ppbv/decade) (1995–2019) to (0.4 [±0.6] ppbv/decade) (1995–2021); and negative trends (high confidence) are found in SON (-0.7 [±0.5] ppbv/decade).
- 5. Nighttime observations measured at MBO show a strong positive median trend (2.0 [±0.6] ppbv/decade) over 2004–2021. A roughly half reduction in magnitude is observed in the 5th percentile after the COVID-19 pandemic (1.9 [±1.4] ppbv/decade). For the 95th percentile, positive trends over 2004–2019 have decreased in 2020 and rebounded in 2021 (4.6 [±1.5] ppbv/decade), likely impacted by wildfires. The trends in all the percentiles show rebounding ozone from 2020 to 2021.
- 6. Twenty-six high-elevation, rural sites were selected from the EPA CASTNET and National Park Service air quality monitoring networks to represent boundary layer variability across the western U.S. Despite sharp ozone reductions in MAM and JJA in 2020, the regional mean, median, 5th and 95th percentile trends changed little and remained strongly negative through the end of 2021 (very high confidence, |SNR| > 3). The 95th percentile trends ($-1.4 [\pm 0.4]$ ppbv/decade) were roughly twice as strong as the trend of the 5th percentile ($-0.6 [\pm 0.3]$ ppbv/decade).



7. Finally, yet importantly, overall opposite percentile and seasonal trends (no overlap of the 2- σ ranges) are found between the free troposphere and the rural surface sites across WNA, except for SON which shows consistent negative trends. By focusing on individual monthly results, we are able to further confirm that the greatest difference occurred in summertime, which suggests that different mechanisms are responsible for increasing ozone in the free troposphere and decreasing ozone in the rural regions.

From 2020 to 2021, even though we found rebounding trends in JJA and continuously diminished trends in DJF/MAM, the long-term mean and median trends show no noticeable change in the free troposphere above WNA. The only structural change we found after the pandemic period is the reduction of trends in the lower percentiles in the California free troposphere and MBO, which do not show evidence of rebound in 2021. In terms of surface ozone at rural sites, albeit with very weak signals, a similar seasonal variability can be detected (rebounding in JJA and declining in DJF/MAM), but the overall mean and median trends remain unchanged.

Based on previous work the most likely explanation for the opposing trends in the free troposphere and at the surface is a sharp contrast in ozone precursor emissions trends in the U.S. boundary layer compared to upwind regions (Fiore et al., 2022). Several regional-scale studies based on observations and chemical transport models have shown that surface ozone is decreasing across most of the United States due to reductions of domestic ozone precursors, primarily NOx (Cooper et al., 2012; Hogrefe et al., 2011; Jaffe et al., 2018; Luo et al., 2020; Simon et al., 2015; Strode et al., 2015). On the hemispheric scale, a strong shift of ozone precursor emissions from high latitudes to low latitudes has increased ozone production in the tropics and subtropics of the Northern Hemisphere and the export of these precursors and resulting ozone to mid-latitudes has contributed to the observed increase of ozone in the free troposphere (Christiansen et al., 2022; Fiore et al., 2022; Gaudel et al., 2020; Y. Zhang et al., 2016, 2021), with additional contributions from the increase in commercial aircraft emissions (Wang et al., 2022), and a small contribution from the global increase of methane (Lin et al., 2017). Our present analysis, which focuses on observed ozone trends, is the first of a series of papers to explore ozone trends and attribution in the free- and lower-troposphere above WNA. We have work in progress to apply numerous satellite products and global atmospheric chemistry models to better characterize the source-attribution and main physiochemical processes driving the observed free- and lower-tropospheric trends.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

The AJAX data can be downloaded at https://doi.org/10.5067/ASDC/AJAX_O3 (Iraci et al., 2021). The IAGOS database is supported in France by AERIS (https://www.aeris-data.fr; Boulanger et al., 2022). Ozonesonde data measured at Boulder, Colorado, and Trinidad Head, California can be downloaded at ftp://aftp.cmdl.noaa.gov/ data/ozwv/Ozonesonde/ (NOAA Global Monitoring Laboratory, 2022a; Sterling et al., 2018). Ozonesonde data measured at Edmonton, Kelowna, and Port Hardy (Canada) can be downloaded at https://hegiftom.meteo.be/ datasets/ozonesondes (Environment and Climate Change Canada, 2022; Tarasick et al., 2016). The TMF lidar data used in this publication are available through the Network for the Detection of Atmospheric Composition Change (NDACC) website: https://www-air.larc.nasa.gov/missions/ndacc/ (McDermid et al., 2002; NASA Jet Propulsion Laboratory, 2022). Mt Bachelor Observatory surface ozone data can be found at https://digital.lib. washington.edu/researchworks/discover?scope=%2F&query=%22mt.+bachelor+observatory%22&submit=&-filtertype_0=title&filter_relational_operator_0=contains&filter_0=data (Mt. Bachelor datasets, 2022). Mauna Loa surface ozone data can be found at https://gml.noaa.gov/aftp/data/ozwv/SurfaceOzone/ (NOAA Global Monitoring Laboratory, 2022b). Surface ozone measurements from EPA CASTNET sites can be found at https:// aqs.epa.gov/aqsweb/airdata/download_files.html (U.S. Environmental Protection Agency Clean Air Markets Division, 2022).



M. G., Bre ents durin . (2022) from it=&filter uulze, N. (vironmen.

Acknowledgments

This research was supported in

part by NOAA cooperative agreements NA17OAR4320101 and NA22OAR4320151, and the NASA Atmospheric Composition Campaign Data Analysis and Modeling (20-ACCDAM20-0083) program. The NASA AJAX project recognizes support from Ames Research Center Director's funds, the NASA Postdoctoral Program. the NASA OCO-2 Science Team, and the California Air Resources Board (Contract No. 17RD004), as well as by the Atmospheric Composition Program through the NASA Internal Scientist Funding Model. The authors acknowledge the strong support of the European Commission. Airbus, and the airlines (Lufthansa, Air France, Austrian, Air Namibia, Cathay Pacific, Iberia and China Airlines, and Hawaiian Airlines) that have carried the MOZAIC or IAGOS equipment and performed the maintenance since 1994. Part of this research was carried out at the Jet Propulsion Laboratory, California Institute of Technology under a contract with the National Aeronautics and Space Administration (80NM0018D004). The Mt. Bachelor Observatory is supported by the National Science Foundation (Grant AGS-1447832) and the National Oceanic and Atmospheric Administration (contract #RA-133R-16-SE-0758).

References

- Asman, W. A. H., Lawrence, M. G., Brenninkmeijer, C. A. M., Crutzen, P. J., Cuijpers, J. W. M., & Nédélec, P. (2003). Rarity of upper-tropospheric low O₃ mixing ratio events during MOZAIC flights. *Atmospheric Chemistry and Physics*, 3(5), 1541–1549. https://doi.org/10.5194/ acp-3-1541-2003
- Bachelor Data Sets, Mt. (2022). Mt. Bachelor data sets [Dataset]. Research Works at the University of Washington. Retrieved from https://digital.lib.washington.edu/researchworks/discover?scope=%2F&query=%22mt.+bachelo r+observatory%22&submit=&filtertype_0=title&filter_relational_operator_0=contains&filter_0=data
- Baur, D., Saisana, M., & Schulze, N. (2004). Modeling the effects of meteorological variables on ozone concentration—A quantile regression approach. *Atmospheric Environment*, 38(28), 4689–4699. https://doi.org/10.1016/j.atmosenv.2004.05.028
- Baylon, P., Jaffe, D. A., Wigder, N., Gao, H., & Hee, J. (2015). Ozone enhancement in western U.S. wildfire plumes at the Mt. Bachelor Observatory: The role of NOx. Atmospheric Environment, 109, 297–304. https://doi.org/10.1016/j.atmosenv.2014.09.013
- Bernays, N., Jaffe, D. A., Petropavlovskikh, I., & Effertz, P. (2022). Comment on "comparison of ozone measurement methods in biomass burning smoke: An evaluation under field and laboratory conditions" by Long et al. (2021). Atmospheric Measurement Techniques, 15(10), 3189–3192. https://doi.org/10.5194/amt-15-3189-2022
- Berrocal, V. J., Gelfand, A. E., & Holland, D. M. (2014). Assessing exceedance of ozone standards: A space-time downscaler for fourth highest ozone concentrations. *Environmetrics*, 25(4), 279–291. https://doi.org/10.1002/env.2273
- Blot, R., Nedelec, P., Boulanger, D., Wolff, P., Sauvage, B., Cousin, J.-M., et al. (2021). Internal consistency of the IAGOS ozone and carbon monoxide measurements for the last 25 yr. *Atmospheric Measurement Techniques*, 14(5), 3935–3951. https://doi.org/10.5194/amt-14-3935-2021
- Bouarar, I., Gaubert, B., Brasseur, G. P., Steinbrecht, W., Doumbia, T., Tilmes, S., et al. (2021). Ozone anomalies in the free troposphere during the COVID-19 pandemic. *Geophysical Research Letters*, 48(16), e2021GL094204. https://doi.org/10.1029/2021GL094204
- Boulanger, D., Thouret, V., & Petzold, A. (2022). In-service Aircraft for a Global Observing System (IAGOS) Data Portal [Dataset]. Aeris. https://doi.org/10.25326/20
- Brauer, M., Freedman, G., Frostad, J., Van Donkelaar, A., Martin, R. V., Dentener, F., et al. (2016). Ambient air pollution exposure estimation for the global burden of disease 2013. *Environmental Science & Technology*, 50(1), 79–88. https://doi.org/10.1021/acs.est.5b03709
- Canty, A. J. (2002). Resampling methods in R: The boot package. The Newsletter of the R Project Volume, 2(3), 2–7.
- Chang, K.-L., Cooper, O. R., Gaudel, A., Allaart, M., Ancellet, G., Clark, H., et al. (2022). Impact of the COVID-19 economic downturn on tropospheric ozone trends: An uncertainty weighted data synthesis for quantifying regional anomalies above western North America and Europe. AGU Advances, 3(2). https://doi.org/10.1029/2021AV000542
- Chang, K.-L., Cooper, O. R., Gaudel, A., Petropavlovskikh, I., & Thouret, V. (2020). Statistical regularization for trend detection: An integrated approach for detecting long-term trends from sparse tropospheric ozone profiles. *Atmospheric Chemistry and Physics*, 20(16), 9915–9938. https://doi.org/10.5194/acp-20-9915-2020
- Chang, K.-L., Petropavlovskikh, I., Cooper, O. R., Schultz, M. G., & Wang, T. (2017). Regional trend analysis of surface ozone observations from monitoring networks in eastern North America, Europe, and East Asia. *Elementa: Science of the Anthropocene*, 5, 50. https://doi.org/10.1525/ elementa.243
- Chang, K.-L., Schultz, M. G., Lan, X., McClure-Begley, A., Petropavlovskikh, I., Xu, X., & Ziemke, J. R. (2021). Trend detection of atmospheric time series: Incorporating appropriate uncertainty estimates and handling extreme events. *Elementa: Science of the Anthropocene*, 9(1), 00035. https://doi.org/10.1525/elementa.2021.00035
- Chouza, F., Leblanc, T., Brewer, M., & Wang, P. (2019). Upgrade and automation of the JPL Table Mountain Facility tropospheric ozone lidar (TMTOL) for near-ground ozone profiling and satellite validation. Atmospheric Measurement Techniques, 12(1), 569–583. https://doi.org/10.5194/ami-12-569-2019
- Christiansen, A., Mickley, L. J., Liu, J., Oman, L. D., & Hu, L. (2022). Multidecadal increases in global tropospheric ozone derived from ozonesonde and surface site observations: Can models reproduce ozone trends? *Atmospheric Chemistry and Physics*, 22(22), 14751–14782. https:// doi.org/10.5194/acp-22-14751-2022
- Clark, H., Bennouna, Y., Tsivlidou, M., Wolff, P., Sauvage, B., Barret, B., et al. (2021). The effects of the COVID-19 lockdowns on the composition of the troposphere as seen by In-service Aircraft for a Global Observing System (IAGOS) at Frankfurt. Atmospheric Chemistry and Physics, 21(21), 16237–16256. https://doi.org/10.5194/acp-21-16237-2021
- Cleveland, W. S. (1979). Robust locally weighted regression and smoothing scatterplots. *Journal of the American Statistical Association*, 74(368), 829–836. https://doi.org/10.1080/01621459.1979.10481038
- Cooper, O. R., Gao, R.-S., Tarasick, D., Leblanc, T., & Sweeney, C. (2012). Long-term ozone trends at rural ozone monitoring sites across the United States, 1990–2010. Journal of Geophysical Research: Atmospheres, 117(D22). https://doi.org/10.1029/2012JD018261
- Cooper, O. R., et al. (Eds.). (2021). WMO Air Quality and Climate Bulletin, No. 1, 3 September 2021. Retrieved from https://library.wmo.int/ index.php?lvl=notice_display&id=21942#,YTIzN99MG70
- Cooper, O. R., Parrish, D. D., Stohl, A., Trainer, M., Nédélec, P., Thouret, V., et al. (2010). Increasing springtime ozone mixing ratios in the free troposphere over western North America. *Nature*, 463(7279), 344–348. https://doi.org/10.1038/nature08708
- Cooper, O. R., Parrish, D. D., Ziemke, J., Balashov, N. V., Cupeiro, M., Galbally, I. E., et al. (2014). Global distribution and trends of tropospheric ozone: An observation-based review. *Elementa: Science of the Anthropocene*, 2. https://doi.org/10.12952/journal.elementa.000029
- Cooper, O. R., Schultz, M. G., Schröder, S., Chang, K.-L., Gaudel, A., Benitez, G. C., et al. (2020). Multi-decadal surface ozone trends at globally distributed remote locations. *Elementa: Science of the Anthropocene*, 8(23). https://doi.org/10.1525/elementa.420
- Davies, W. E., Vaughan, G., & O'Connor, F. M. (1998). Observation of near-zero ozone concentrations in the upper troposphere at mid-latitudes. *Geophysical Research Letters*, 25(8), 1173–1176. https://doi.org/10.1029/98GL00909
- Dunn, R. J. H., Aldred, F., Gobron, N., Miller, J. B., Willett, K. M., Ades, M., et al. (2022). Global climate. In State of the climate in 2021 (pp. S11–S142). Bulletin of the American Meteorological Society. https://doi.org/10.1175/BAMS-D-22-0092.1
- Eastoe, E. F., & Tawn, J. A. (2009). Modeling non-stationary extremes with application to surface-level ozone. Journal of the Royal Statistical Society: Series C (Applied Statistics), 58(1), 25–45. https://doi.org/10.1111/j.1467-9876.2008.00638.x
- Emberson, L. D., Ashmore, M. R., Cambridge, H. M., Simpson, D., & Tuovinen, J.-P. (2000). Modeling stomatal ozone flux across Europe. *Environmental Pollution*, 109(3), 403–413. https://doi.org/10.1016/S0269-7491(00)00043
- Environment and Climate Change Canada. (2022). Canadian ozonesonde network [Dataset]. World Ozone and Ultraviolet Radiation Data Centre. Retrieved from https://hegiftom.meteo.be/datasets/ozonesondes
- Fasiolo, M., Wood, S. N., Zaffran, M., Nedellec, R., & Goude, Y. (2020). Fast calibrated additive quantile regression. Journal of the American Statistical Association, 116(535), 1–11. https://doi.org/10.1080/01621459.2020.1725521



- Filonchyk, M., Peterson, M. P., & Sun, D. (2022). Deterioration of air quality associated with the 2020 U.S. wildfires. Science of the Total Environment, 826, 154103. https://doi.org/10.1016/j.scitotenv.2022.154103
- Fiore, A. M., Hancock, S., Lamarque, J.-F., Correa, G., Chang, K.-L., Ru, M., et al. (2022). Understanding recent tropospheric ozone trends in the context of large internal variability: A new perspective from chemistry-climate model ensembles. *Environmental Research: Climate*, 1(2), 025008. https://doi.org/10.1088/2752-5295/ac9cc2
- Fischer, E. V., Jaffe, D. A., & Weatherhead, E. C. (2011). Free tropospheric peroxyacetyl nitrate (PAN) and ozone at Mount Bachelor: Potential causes of variability and timescale for trend detection. *Atmospheric Chemistry and Physics*, 11(12), 5641–5654. https://doi.org/10.5194/acp-11-5641-2011
- Fleming, Z. L., Doherty, R. M., von Schneidemesser, E., Malley, C. S., Cooper, O. R., Pinto, J. P., et al. (2018). Tropospheric Ozone Assessment Report: Present-day ozone distribution and trends relevant to human health. *Elementa: Science of the Anthropocene*, 6, 12. https://doi. org/10.1525/elementa.291
- Gaudel, A., Cooper, O. R., Chang, K.-L., Bourgeois, I., Ziemke, J. R., Strode, S. A., et al. (2020). Aircraft observations since the 1990s reveal increases of tropospheric ozone at multiple locations across the Northern Hemisphere. *Science Advance*, 6(34), eaba8272. https://doi.org/10.1126/sciady.aba8272
- Gelaro, R., McCarty, W., Suárez, M. J., Todling, R., Molod, A., Takacs, L., et al. (2017). The Modern-Era Retrospective Analysis for Research and Applications, Version 2 (MERRA-2). Journal of Climate, 30(14), 5419–5454. https://doi.org/10.1175/JCLI-D-16-0758.1
- Gilleland, E. (2020). Bootstrap methods for statistical inference. Part II: Extreme-value analysis. Journal of Atmospheric and Oceanic Technology, 37(11), 2135–2144. https://doi.org/10.1175/JTECH-D-20-0070.1
- Gkatzelis, G. I., Gilman, J. B., Brown, S. S., Eskes, H., Gomes, A. R., Lange, A. C., et al. (2021). The global impacts of COVID-19 lockdowns on urban air pollution: A critical review and recommendations. *Elementa: Science of the Anthropocene*, 9(1). https://doi.org/10.1525/ elementa.2021.00176
- Gonçalves, S., & White, H. (2005). Bootstrap standard error estimates for linear regression. Journal of the American Statistical Association, 100(471), 970–979. https://doi.org/10.1198/01621450400002087
- Grant, W. B., Browell, E. V., Butler, C. F., Fenn, M. A., Clayton, M. B., Hannan, J. R., et al. (2000). A case study of transport of tropical marine boundary layer and lower tropospheric air masses to the northern midlatitude upper troposphere. *Journal of Geophysical Research: Atmos*pheres, 105(D3), 3757–3769. https://doi.org/10.1029/1999JD901022
- Gratz, L. E., Jaffe, D. A., & Hee, J. R. (2015). Causes of increasing ozone and decreasing carbon monoxide in springtime at the Mt. Bachelor Observatory from 2004 to 2013. Atmospheric Environment, 109, 323–330. https://doi.org/10.1016/j.atmosenv.2014.05.076
- Gulev, S. K., Thorne, P. W., Ahn, J., Dentener, F. J., Domingues, C. M., Gerland, S., et al. (2021). Chapter 2: Changing state of the climate system. In V. Masson-Delmotte et al. (Eds.), Climate change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 287–422). Cambridge University Press. https://doi. org/10.1017/9781009157926
- Hastie, T. J., & Tibshirani, R. J. (1990). Generalized additive models (Vol. 43). CRC Press.
- Hogrefe, C., Hao, W., Zalewsky, E. E., Ku, J.-Y., Lynn, B., Rosenzweig, C., et al. (2011). An analysis of long-term regional-scale ozone simulations over the Northeastern United States: Variability and trends. Atmospheric Chemistry and Physics, 11(2), 567–582. https://doi.org/10.5194/ acp-11-567-2011
- Hyndman, R. J., & Fan, Y. (1996). Sample quantiles in statistical packages. The American Statistician, 50(4), 361–365. https://doi.org/10.1080/ 00031305.1996.10473566
- Iraci, L. T., Yates, E. L., Marrero, J. E., Parworth, C. L., Ryoo, J.-M., Tanaka, T., et al. (2021). Airborne Trace Gas Measurements Collected by the Alpha Jet Atmospheric eXperiment (AJAX) Project: 2011–2018 [Dataset]. Atmospheric Science Data Center. https://doi.org/10.5067/ASDC/AJAX_O3
- Jaffe, D. A., Cooper, O. R., Fiore, A. M., Henderson, B. H., Tonnesen, G. S., Russell, A. G., et al. (2018). Scientific assessment of background ozone over the U.S.: Implications for air quality management. *Elementa: Science of the Anthropocene*, 6. https://doi.org/10.1525/elementa.309
- Jaffe, D. A., Ninneman, M., & Chan, H. C. (2022). NOx and O₃ Trends at U.S. non-attainment areas for 1995–2020: Influence of COVID-19 reductions and wildland fires on policy-relevant concentrations. *Journal of Geophysical Research: Atmospheres*, 127(11), e2021JD036385. https://doi.org/10.1029/2021JD036385
- Jaffe, D. A., & Zhang, L. (2017). Meteorological anomalies lead to elevated O₃ in the western U.S. in June 2015. Geophysical Research Letters, 44(4), 1990–1997. https://doi.org/10.1002/2016GL072010
- Keller, C. A., Evans, M. J., Knowland, K. E., Hasenkopf, C. A., Modekurty, S., Lucchesi, R. A., et al. (2021). Global impact of COVID-19 restrictions on the surface concentrations of nitrogen dioxide and ozone. *Atmospheric Chemistry and Physics*, 21(5), 3555–3592. https://doi. org/10.5194/acp-21-3555-2021
- Koenker, R. (1994). Confidence intervals for regression quantiles. In P. Mandl, & M. Hušková (Eds.), Asymptotic statistics (pp. 349–359). Springer.
- Koenker, R., & Hallock, K. F. (2001). Quantile regression. The Journal of Economic Perspectives, 15(4), 143–156. https://doi.org/10.1257/jep.15.4.143
- Koumoutsaris, S., Bey, I., Generoso, S., & Thouret, V. (2008). Influence of El Niño–Southern Oscillation on the interannual variability of tropospheric ozone in the northern mid-latitudes. *Journal of Geophysical Research: Atmospheres*, 113(D19), D19301. https://doi. org/10.1029/2007JD009753
- Krock, M., Bessac, J., Stein, M. L., & Monahan, A. H. (2022). Nonstationary seasonal model for daily mean temperature distribution bridging bulk and tails. Weather and Climate Extremes, 36, 100438. https://doi.org/10.1016/j.wace.2022.100438
- Kunsch, H. R. (1989). The jackknife and the bootstrap for general stationary observations. Annals of Statistics, 17(3), 1217–1241. https://doi.org/10.1214/aos/1176347265
- Kyselý, J. (2008). A cautionary note on the use of nonparametric bootstrap for estimating uncertainties in extreme-value models. Journal of Applied Meteorology and Climatology, 47(12), 3236–3251. https://doi.org/10.1175/2008JAMC1763.1
- Laing, J. R., Jaffe, D. A., & Hee, J. R. (2016). Physical and optical properties of aged biomass burning aerosol from wildfires in Siberia and the Western USA at the Mt. Bachelor Observatory. *Atmospheric Chemistry and Physics*, 16(23), 15185–15197. https://doi.org/10.5194/ acp-16-15185-2016
- Langford, A. O. (1999). Stratosphere-troposphere exchange at the subtropical jet: Contribution to the tropospheric ozone budget at mid-latitudes. *Geophysical Research Letters*, 26(16), 2449–2452. https://doi.org/10.1029/1999g1900556
- Lin, M., Horowitz, L. W., Cooper, O. R., Tarasick, D., Conley, S., Iraci, L. T., et al. (2015). Revisiting the evidence of increasing springtime ozone mixing ratios in the free troposphere over western North America. *Geophysical Research Letters*, 42(20), 8719–8728. https://doi. org/10.1002/2015GL065311



- Lin, M., Horowitz, L. W., Payton, R., Fiore, A. M., & Tonnesen, G. (2017). U.S. surface ozone trends and extremes from 1980 to 2014: Quantifying the roles of rising Asian emissions, domestic controls, wildfires, and climate. *Atmospheric Chemistry and Physics*, 17(4), 2943–2970. https://doi.org/10.5194/acp-17-2943-2017
- Liu, G., Tarasick, D. W., Fioletov, V. E., Sioris, C. E., & Rochon, Y. J. (2009). Ozone correlation lengths and measurement uncertainties from analysis of historical ozonesonde data in North America and Europe. *Journal of Geophysical Research: Atmospheres*, 114(D4), D04112. https://doi.org/10.1029/2008JD010576
- Luo, H., Astitha, M., Rao, S. T., Hogrefe, C., & Mathur, R. (2020). Assessing the manageable portion of ground-level ozone in the contiguous United States. Journal of the Air & Waste Management Association, 70(11), 1136–1147. https://doi.org/10.1080/10962247.2020.1805375
- Malashock, D. A., DeLang, M. N., Becker, J. S., Serre, M. L., West, J. J., Chang, K.-L., et al. (2022). Estimates of ozone concentrations and attributable mortality in urban, peri-urban, and rural areas worldwide in 2019. *Environmental Research Letters*, 17(5), 054023. https://doi. org/10.1088/1748-9326/ac66f3
- McDermid, I. S., Beyerle, G., Haner, D. A., & Leblanc, T. (2002). Redesign and improved performance of the tropospheric ozone lidar at the Jet Propulsion Laboratory Table Mountain Facility. *Applied Optics*, *41*(36), 7550–7555. https://doi.org/10.1364/AO.41.007550
- Michael, R., O'Lenick, C. R., Monaghan, A., Wilhelmi, O., Wiedinmyer, C., Hayden, M., & Estes, M. (2019). Application of geostatistical approaches to predict the spatio-temporal distribution of summer ozone in Houston, Texas. *Journal of Exposure Science and Environmental Epidemiology*, 29(6), 806–820. https://doi.org/10.1038/s41370-018-0091-4
- Mills, G., Hayes, F., Simpson, D., Emberson, L., Norris, D., Harmens, H., & Bueker, P. (2011). Evidence of widespread effects of ozone on crops and (semi-) natural vegetation in Europe (1990–2006) in relation to AOT40- and flux-based risk maps. *Global Change Biology*, 17(1), 592–613. https://doi.org/10.1111/j.1365-2486.2010.02217.x
- Mills, G., Pleijel, H., Malley, C. S., Sinha, B., Cooper, O. R., Schultz, M. G., et al. (2018). Tropospheric Ozone Assessment Report: Present-day tropospheric ozone distribution and trends relevant to vegetation. *Elementa: Science of the Anthropocene*, 6, 47. https://doi.org/10.1525/ elementa.302
- Miyazaki, K., Bowman, K., Sekiya, T., Takigawa, M., Neu, J. L., Sudo, K., et al. (2021). Global tropospheric ozone responses to reduced NOx emissions linked to the COVID-19 worldwide lockdowns. *Science Advance*, 7(24), eabf7460. https://doi.org/10.1126/sciadv.abf7460
- Munir, S., Chen, H., & Ropkins, K. (2012). Modeling the impact of road traffic on ground-level ozone concentration using a quantile regression approach. Atmospheric Environment, 60, 283–291. https://doi.org/10.1016/j.atmosenv.2012.06.043
- NASA Jet Propulsion Laboratory. (2022). Table Mountain atmospheric lidar data [Dataset]. Network for the Detection of Atmospheric Composition Change. Retrieved from https://www-air.larc.nasa.gov/missions/ndacc/
- Nédélec, P., Blot, R., Boulanger, D., Athier, G., Cousin, J.-M., Gautron, B., et al. (2015). Instrumentation on commercial aircraft for monitoring the atmospheric composition on a global scale: The IAGOS system, technical overview of ozone and carbon monoxide measurements. *Tellus* B: Chemical and Physical Meteorology, 67(1), 27791. https://doi.org/10.3402/tellusb.v67.27791
- Neu, J. L., Flury, T., Manney, G. L., Santee, M. L., Livesey, N. J., & Worden, J. (2014). Tropospheric ozone variations governed by changes in stratospheric circulation. *Nature Geoscience*, 7(5), 340–344. https://doi.org/10.1038/ngeo2138
- NOAA Global Monitoring Laboratory. (2022a). Homogenized ozonesonde data archive [Dataset]. Retrieved from ftp://aftp.cmdl.noaa.gov/data/ ozwv/Ozonesonde/
- NOAA Global Monitoring Laboratory. (2022b). Mauna Loa surface ozone data [Dataset]. Retrieved from https://gml.noaa.gov/aftp/data/ozwv/ SurfaceOzone/
- Nowak, J. B., Parrish, D. D., Neuman, J. A., Holloway, J. S., Cooper, O. R., Trainer, M., et al. (2004). Gas-phase chemical characteristics of Asian emission plumes observed during ITCT 2K2 over the eastern North Pacific Ocean. *Journal of Geophysical Research: Atmospheres*, 109(D23). https://doi.org/10.1029/2003JD004488
- Oltmans, S. J., Lefohn, A. S., Harris, J. M., Galbally, I., Scheel, H. E., Bodeker, G., et al. (2006). Long-term changes in tropospheric ozone. *Atmospheric Environment*, 40(17), 3156–3173. https://doi.org/10.1016/j.atmosenv.2006.01.029
- Oltmans, S. J., Lefohn, A. S., Shadwick, D., Harris, J. M., Scheel, H. E., Galbally, I., et al. (2013). Recent tropospheric ozone changes—A pattern dominated by slow or no growth. Atmospheric Environment, 67, 331–351. https://doi.org/10.1016/j.atmosenv.2012.10.057
- Petzold, A., Thouret, V., Gerbig, C., Zahn, A., Brenninkmeijer, C. A., Gallagher, M., et al. (2015). Global-scale atmosphere monitoring by in-service aircraft-current achievements and future prospects of the European research infrastructure IAGOS. *Tellus B: Chemical and Physical Meteorology*, 67(1), 28452. https://doi.org/10.3402/tellusb.v67.28452
- Politis, D. N., Romano, J. P., & Wolf, M. (1997). Subsampling for heteroskedastic time series. Journal of Econometrics, 81(2), 281–317. https:// doi.org/10.1016/S0304-4076(97)86569-4
- Porter, W. C., Heald, C. L., Cooley, D., & Russell, B. (2015). Investigating the observed sensitivities of air-quality extremes to meteorological drivers via quantile regression. Atmospheric Chemistry and Physics, 15(18), 10349–10366. https://doi.org/10.5194/acp-15-10349-2015
- Reich, B. J. (2012). Spatiotemporal quantile regression for detecting distributional changes in environmental processes. Journal of the Royal Statistical Society: Series C (Applied Statistics), 61(4), 535–553. https://doi.org/10.1111/j.1467-9876.2011.01025.x
- Sillman, S. (1999). The relation between ozone, NOx, and hydrocarbons in urban and polluted rural environments. *Atmospheric Environment*, 33(12), 1821–1845. https://doi.org/10.1016/s1352-2310(98)00345-8
- Simon, H., Reff, A., Wells, B., Xing, J., & Frank, N. (2015). Ozone trends across the United States over a period of decreasing NOx and VOC emissions. *Environmental Science & Technology*, 49(1), 186–195. https://doi.org/10.1021/es504514z
- Smith, R. L. (1989). Extreme value analysis of environmental time series: An application to trend detection in ground-level ozone. *Statistical Science*, 4(4), 367–377. https://doi.org/10.1214/ss/1177012400
- Sokhi, R. S., Singh, V., Querol, X., Finardi, S., Targino, A. C., de Fatima Andrade, M., et al. (2021). A global observational analysis to understand changes in air quality during exceptionally low anthropogenic emission conditions. *Environment International*, 157, 106818. https://doi. org/10.1016/j.envint.2021.106818
- Sousa, S., Pires, J., Martins, F., Pereira, M., & Alvim-Ferraz, M. (2009). Potentialities of quantile regression to predict ozone concentrations. *Environmetrics*, 20(2), 147–158. https://doi.org/10.1002/env.916
- Stauffer, R. M., Thompson, A. M., Kollonige, D. E., Tarasick, D. W., VanMalderen, R., Smit, H. G., et al. (2022). An examination of the recent stability of ozonesonde global network data. *Earth and Space Science*, 9(10), e2022EA002459. https://doi.org/10.1029/2022ea002459
- Steinbrecht, W., Kubistin, D., Plass-Dülmer, C., Davies, J., Tarasick, D. W., von der Gathen, P., et al. (2021). COVID-19 crisis reduces free tropospheric ozone across the Northern Hemisphere. *Geophysical Research Letters*, 48(5), e2020GL091987. https://doi.org/10.1029/2020GL091987
- Sterling, C. W., Johnson, B. J., Oltmans, S. J., Smit, H. G. J., Jordan, A. F., Cullis, P. D., et al. (2018). Homogenizing and estimating the uncertainty in NOAA's long-term vertical ozone profile records measured with the electrochemical concentration cell ozonesonde. Atmospheric Measurement Techniques, 11(6), 3661–3687. https://doi.org/10.5194/amt-11-3661-2018



- Stohl, A., Bonasoni, P., Cristofanelli, P., Collins, W., Feichter, J., Frank, A., & Kentarchos, T. (2003). Stratosphere-troposphere exchange: A review, and what we have learned from STACCATO. *Journal of Geophysical Research: Atmospheres*, 108(D12), 8516. https://doi. org/10.1029/2002JD002490
- Strode, S. A., Rodriguez, J. M., Logan, J. A., Cooper, O. R., Witte, J. C., Lamsal, L. N., et al. (2015). Trends and variability in surface ozone over the United States. *Journal of Geophysical Research: Atmospheres*, 120(17), 9020–9042. https://doi.org/10.1002/2014JD022784
- Szopa, S., Naik, V., Adhikary, B., Artaxo, P., Berntsen, T., Collins, W. D., et al. (2021). Chapter 6: Short-lived climate forcers. In V. Masson-Delmotte, et al. (Eds.), Climate change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 817–922). Cambridge University Press. https://doi.org/10.1017/9781009157926.008
- Tarasick, D. W., Davies, J., Smit, H. G. J., & Oltmans, S. J. (2016). A re-evaluated Canadian ozonesonde record: Measurements of the vertical distribution of ozone over Canada from 1966 to 2013. Atmospheric Measurement Techniques, 9(1), 195–214. https://doi.org/10.5194/ amt-9-195-2016
- Tarasick, D. W., Galbally, I. E., Cooper, O. R., Schultz, M. G., Ancellet, G., Leblanc, T., et al. (2019). Tropospheric Ozone Assessment Report: Tropospheric ozone from 1877 to 2016, observed levels, trends, and uncertainties. *Elementa: Science of the Anthropocene*, 7(39). https://doi. org/10.1525/elementa.376
- U.S. Environmental Protection Agency Clean Air Markets Division. (2022). Clean Air Status and Trends Network (CASTNET) [Dataset]. Retrieved from https://aqs.epa.gov/aqsweb/airdata/download_files.html
- Wang, H., Lu, X., Jacob, D. J., Cooper, O. R., Chang, K.-L., Li, K., et al. (2022). Global tropospheric ozone trends, attributions, and radiative impacts in 1995–2017: An integrated analysis using aircraft (IAGOS) observations, ozonesonde, and multi-decadal chemical model simulations. Atmospheric Chemistry and Physics, 22(20), 13753–13782. https://doi.org/10.5194/acp-22-13753-2022
- Weatherhead, E. C., Reinsel, G. C., Tiao, G. C., Meng, X.-L., Choi, D., Cheang, W.-K., et al. (1998). Factors affecting the detection of trends: Statistical considerations and applications to environmental data. *Journal of Geophysical Research: Atmospheres*, 103(D14), 17149–17161. https://doi.org/10.1029/98JD00995
- Wood, S. N. (2006). Generalized additive models: An introduction with R. CRC Press.
- Wood, S. N., Li, Z., Shaddick, G., & Augustin, N. H. (2017). Generalized additive models for gigadata: Modeling the UK black smoke network daily data. Journal of the American Statistical Association, 112(519), 1199–1210. https://doi.org/10.1080/01621459.2016.1195744
- Xu, B., & Lin, B. (2018). What cause large regional differences in PM₂₅ pollutions in China? Evidence from quantile regression model. Journal of Cleaner Production, 174, 447–461. https://doi.org/10.1016/j.jclepro.2017.11.008
- Yates, E. L., Iraci, L. T., Singh, H. B., Tanaka, T., Roby, M., Hamill, P., et al. (2016). Airborne measurements and emission estimates of greenhouse gases and other trace constituents from the 2013 California Yosemite Rim wildfire. *Atmospheric Environment*, 127, 293–302. https:// doi.org/10.1016/j.atmosenv.2015.12.038
- Yates, E. L., Johnson, M. S., Iraci, L. T., Ryoo, J.-M., Pierce, R. B., Cullis, P. D., et al. (2017). An assessment of ground level and free tropospheric ozone over California and Nevada. *Journal of Geophysical Research: Atmospheres*, 122(18), 10089–10102. https://doi. org/10.1002/2016JD026266
- Zhang, L., & Jaffe, D. A. (2017). Trends and sources of ozone and sub-micron aerosols at the Mt. Bachelor Observatory (MBO) during 2004– 2015. Atmospheric Environment, 165, 143–154. https://doi.org/10.1016/j.atmosenv.2017.06.042
- Zhang, Y., Cooper, O. R., Gaudel, A., Thompson, A. M., Nédélec, P., Ogino, S.-Y., & West, J. J. (2016). Tropospheric ozone change from 1980 to 2010 dominated by equatorward redistribution of emissions. *Nature Geoscience*, 9(12), 875–879. https://doi.org/10.1038/ngeo2827
- Zhang, Y., West, J. J., Emmons, L. K., Flemming, J., Jonson, J. E., Lund, M. T., et al. (2021). Contributions of world regions to the global tropospheric ozone burden change from 1980 to 2010. Geophysical Research Letters, 48(1), e2020GL089184. https://doi.org/10.1029/2020GL089184
- Ziemke, J. R., Kramarova, N. A., Frith, S. M., Huang, L.-K., Haffner, D. P., Wargan, K., et al. (2022). NASA satellite measurements show global-scale reductions in free tropospheric ozone in 2020 and again in 2021 during COVID-19. *Geophysical Research Letters*, 49(15), e2022GL098712. https://doi.org/10.1029/2022GL098712
- Ziemke, J. R., Oman, L. D., Strode, S. A., Douglass, A. R., Olsen, M. A., McPeters, R. D., et al. (2019). Trends in global tropospheric ozone inferred from a composite record of TOMS/OMI/MLS/OMPS satellite measurements and the MERRA-2 GMI simulation. Atmospheric Chemistry and Physics, 19(5), 3257–3269. https://doi.org/10.5194/acp-19-3257-2019